ļ	PLANT		STIPULATED		
	ACCOUNT	PLANT ACCOUNT DESCRIPTION	DEPR RATE		
	GENERAL PLAN	NT			
	39000	Gen-Structures & Improvements-Elec	2.12%		
	39000	Gen-Structures & Improvements-Elec - Wolf Creek	3.83%		
	39000	Gen-Structures & Improvements-Elec - Composite Rate	2.14%		
	39100	Gen-Office Furniture & Equip-Elcc	4.00%	Gen Plt Amort	25 yr
	39101	Gen-Office Furniture & Equip- Elec - Wolf Creek	4.00%	Gen Plt Amort	25 yr
	39102	Gen-Office Furniture-Computer	20,00%	Gen Plt Amort	5 ут
	39200	Gen-Transportation Equip- Autos -Elec	9.37%		
	39201	Gen-Transportation Equip- Light Trucks -Elec	10.75%		
	39202	Gen-Transportation Equip- Heavy Trucks -Elec	9.61%		
	39203	Gen-Transportation Equip- Tractors -Elec	7.84%		
	39204	Gen-Transportation Equip- Trailers-Elec	3.82%		
	39300	Gen-Stores Equip-Elec	4,00%	Gen Plt Amort	25 yr
	39400	Gen-Tools, Shop and Garage Equip-Elec	4.00%	Gen Plt Amort	25 yr
	39500	Gen-Laboratory Equip-Elec	4.00%	Gen Plt Amort	25 уг
	39600	Gen-Power Operated Equip-Elec	4.83%		-
	39700	Gen-Communication Equip-Elec	6.67%	Gen Plt Amort	15 yr
	39701	Gen-Communication Equip-Elec - Wolf Creek	6.67%	Gen Plt Amort	15 yr
	39800	Gen-Misc Equip	6.67%	Gen Plt Amort	15 yr

#### Security Tracker Cost Definition

The Security Tracker is for incremental costs spent to meet continuously emerging security threats to critical infrastructure and growing regulatory requirements for protection of critical infrastructure, inclusive of Department of Defense ("DOD"), Department of Homeland Security ("DHS"), Department of Energy ("DOE"), Nuclear Regulatory Commission ("NRC"), Securities and Exchange Commission ("SEC"), Federal Communications Commission ("FCC"), Federal Energy Regulatory Commission ("FERC"), North American Electric Reliability Corporation ("NERC"), etc., or security needs. Historically, the impacts to Evergy have been heavily focused on cybersecurity and the growing attack surface in cyber warfare that require the critical infrastructure industries to invest in security to protect the electric system. Today, the threats to critical infrastructure persist and continue to grow inclusive of physical security. These regulatory obligations, such as NERC Critical Infrastructure Protection ("CIP") Standards, are publicly available and subject to federal audits. Security needs are driven by many government entities, threat intelligence and analytics as well as industry best practices.

#### Non-Labor O&M Calculation Tracker Baseline EKM:

The specific CIP / Cybersecurity Tracker baseline amount is set at the sum of the costs from July 1, 2022, to June 30, 2023. amounting to \$4,184,570 on a total Evergy Metro basis. This baseline amount of non-labor costs is considered included in EKM's s retail revenue requirement resulting from this rate proceeding, the 23-775 Docket. All non-labor CIP / Cybersecurity compliance costs identified as the Company continues to incur costs for protection of its critical infrastructure assets will be tracked against this baseline amount.

Regulatory Asset/Regulatory Liability: Actual CIP / Cybersecurity costs incurred for the 12-month period beginning with the first day of the month closest to the effective date of rates in this 23-775 Docket through the calendar year from that day, and each 12-month period beginning thereafter, will be compared to the baseline cost amount identified above. Such costs will be supported by vendor invoices. If the 12-month period cost is in excess of the baseline cost, then a regulatory asset will be established. If the 12-month period cost is below the baseline cost, then a regulatory liability will be established. In the event that a subsequent full general rate case update period occurs prior to the end of a 12-month tracking period, the baseline costs will be compared to the actual costs and a regulatory asset or regulatory liability will be established using the principles described above. These regulatory assets and/or liabilities will then be considered for recovery through amortization to cost of service in the Company's next general rate case.

#### **Sunset Provision**

The Security Tracker will terminate upon completion of the first Evergy full general rate proceeding filed on or after January 1, 2028. If Evergy wishes to continue the Security Tracker beyond that time, Evergy must propose such action to the Commission. In that proceeding, Evergy may request the Security Tracker mechanism be reauthorized and continued. Evergy will bear the burden of showing the extension of the Security Tracker is in the public interest and will result in just and reasonable rates. All other parties retain the right to object to an extension of the Security Tracker in that future proceeding.

.

#### EKM Regulatory Assets & Liabilities Docket No. 23-EKCE-775-RTS

Adjustment	Asset/Liability	Description	Total Balance To Be Amortized	Period (Year)	Start Date	End Date	Annual Amortization	True-Up Balance @6/30/23	
Removed from	Rémoved from revenue requirement								
CS-80	Asset	2018 Rate Case Expenses	632,272	5	01/2019	12/2023	0	63,227	
CS-88	Liability	CIPS/Cybersecurity Tracker	(2,182,363)	5	01/2019	12/2023	0	(218,236)	
CS-110	Asset	2011 Flood Expense	924,928	10	01/2013	12/2022	0	0	
CS-130	Asset	Customer Migration	191,125	5	01/2019	12/2023	0	19,113	
CS-131	Liability	LaCygne Budgeted Plant	(1,985.865)	5	01/2019	12/2023	0	(198,597)	
CS-132	Liability	LaCygne Depreciation Deferral	(11,666)	5	01/2019	12/2023	0	(1,167)	
CS-133	Liability	Wolf Creek Budgeted Plant	(83,350)	5	01/2019	12/2023	0	(8,335)	
Continued from	n prior rate case								
RB-25/CS-111 RB-25/CS-111	Asset Asset	latan 1 and Common Vintage 1 Jatan 1 and Common Vintage 2	2,855,631 631,250	47 44.9	12/2010 01/2013	11/2057 11/2057	60,758 14,059	2,091,091 483,631	
RB-27/CS-113	Asset	La Cygne Depreciation Deferral	2,957,141	25	10/2015	09/2040	118,286	2,040,427	
RB-55	Liability	Emission Allowance	(36,987,232)	22	12/2010	11/2032	(1,681,238)	(15,810,094)	
CS-95	Asset	Merger Transition Costs	7,692,018	10	01/2019	12/2028	769,202	4,230,610	
Proposed in cu	rrent rate case								
CS-29	Asset	COVID AAO Expenses	2,379,060	3	01/2024	12/2026	793,020	2,379,060	(A)
CS-80	Asset	2023 Rate Case Expenses	1,794,495	3	01/2024	12/2026	598,165	1,794,495	(A)
CS-88	Liability	CIPS/Cybersecurity Tracker	(8,154,199)	3	01/2024	12/2026	(2,718,066)	(8,154,199)	
CS-118	Asset	Unrecovered Reserve - Meter Replacement	2,509,511	3	01/2024	12/2026	836,504	2,509,511	
CS-134	Asset	Loss Revenue - TOU & RD	121,655	3	01/2024	12/2026	40,552	121,655	
CS-135	Asset	TOU, RD, and Residential DG Costs	1,719,199	3	01/2024	12/2026	573,066	1,719,199	
CS-137	Liability	Environmental Insurance Settlements	(3,084,772)	3	01/2024	12/2026	(1,028,257)	(3,084,772)	
CS-138	Asset	Electrification	256,545	3	01/2024	12/2026	85,515	501,997	

(A) Not included in the regulatory asset/liability tracking

#### EXHIBIT EKM - 3

#### Evergy Kansas Metro - Bright Lines Mitigation Population - Account #/Service Agreement ID#

1791460348-1798648902 8088021143-8080581135 1896316213-1891958645 4043833910-4045920863 4026254198-4029536046 2134496398-2131963292 1511115882-1515074212 2074994508-2070619217 3552500186-3550334931 4856377067-4851743273 4548696184-4548056782 0929092170-0922337985 3715528779-3719378880 3806879855-3801152040 2664962362-2661670310 2629684550-2621288407 2050505454-2050627334 4265540612-4269155950 9560627364-9560899698 3047943455-3042010240 7428859305-7420826217 1579556412-1572530813 9352955988-9350077496 8106624142-8100290153 1334869038-1331055738 5690817610-5691283735 3579834824-3570829414 1575716921-1578792172 3873831279-3871911435 4319670524-4316187359 1729493808-1720219235 3454948026-3451631213 3576009546-3570564184 7202512986-7201305034 5181847744-5180619065

8655603005-8650877341 0702282196-0700104669 1592821001-1590257950 22183.11299-2211791006 9320086710-9324181514 3987630909-3980486138 8003076282-8000679361 3093079466-3091722325 3679568436-3672872841 2160282586-2160437473 8080192826-8080773640 7850981944-7852001247 1813576755-1810529123 7178015284-7173159816 6069525169-6064553169 1009104315-1004598498 2779169438-2770127556 3597104860-3590840738 3058631382-3050581672 9514864627-9516089167 9884011140-9886704466 6911344397-6911390691 8557672006-8551322664 2105518908-2101562353 7913862647-7911120905 0160377071-0161254536 1698979886-1696198903 9037911112-9033553765 0119927129-0110379575 6574049273-6571741564 7876300797-7871314302 3742402774-3740426211 2898373242-2890814197 1767649547-1760275449 6934757937-6930040590

3182329890-3181009774 0473051688-0471483317 1347517971-1348234904 4241905596-4241301295 0678404360-0670483259 2243628472-2241021844 6516087106-6511791443 8039130004-8031256369 0151227508-0152372600 0247501089-0241559291 5143724894-5140956738 2201215153-2201175155 3659587784-3650419484 5768751306-5761647870 1109134593-1104625203 6579003360-6576885944 7214484576-7211818329 1928612542-1921813940 9077665175-9070828377 6301999981-6300326242 1302102536-1309220393 0779500668-0771225861 4071015080-4071469098 6365521722-6361238043 1022004335-1021639520 4047589757-4040278501 9493194871-9491981138 0087002125-0087190645 6671423648-6671194445 9130278513-9131920188 0052309002-0051698847 0368794464-0360879763 0927270365-0921177336 6059791448-6051101047 0071044016-0077499194

5546074392-5545574021 1401394489-1400717624 2176371541-2171287401 2945207046-2941322201 5085418131-5089758946 2139663517-2131631981 8173050008-8177669601 6923151070-6926092320 9774959742-9773670582 3525926778-3520588724 2732888228-2730167973 5371099894-5375738555 7079488904-7074537794 1175684641-1178410631 7387569222-7380988299 5929043079-5920664935 9938461995-9931659905 2561279646-2561437109 0272923311-0271507009 9331813299-9332441454 8120024798-8128597353 8858283171-8850194812 6174924385-6173793372 9855611253-9852110544 1549253744-1541479823 3716424328-3711420597 0122826370-0128468674 9536383942-9531939146 4010711082-4011571352 9763040478-9760821340 4997419093-4999627019 3442752072-3441654231 5003176716-5000101346 6056843052-6051137116 3180924940-3180298398

## EXHIBIT EKM - 3

# Evergy Kansas Metro – Bright Lines Mitigation Population – Account #/Service Agreement ID#

8751621148-8750439769
1597090392-1590098844
8074753489-8070783057
8980291458-8981773404
9510318979-9511065637
5659347507-5651906746
5023097228-5021236467
9918204146-9910105471
3839405277-3836909834
1334475915-1338578768
6812028243-6811352433
0040030834-0041338967
0066381775-0060464133
4357543992-4351667245
6344785048-6340183510
7611929093-7618960680
4968770467-4961802553
8953932124-8951827564
5188682710-5181183723
0352483826-0350982979
0319210298-0311085311
8111985676-8119912659
9434640074-9431201006
6502556907-6501572984
9654100350-9650640763
6110143355-6110241137
7719861909-7710242950
0842565574-0841805915
9213047978-9211750379
6666172414-6660763085
8788107419-8780011997
9086162680-9080981858
1351273656-1351870021
0579958686-0579857894
9129398674-9120543122

1302102536-13076230	71
5094990416-50910634	63
6625101380-66256827	68
0396224070-03983790	85
2896906844-289131634	41
8497879896-84960598	77
6944004792-69413694	83
9095073041-90903234	90
4333606791-43312316	43
3446609926-34408737:	57
1014510493-10155098	34
7876381213-78713349	92
2083965476-20843141	80
4806629494-48003085	87
4523851113-45243813	74
3671828748-36746482	01
9590111149-95906920	34
9269895560-92609672	14
1325454531-13216395	83
8846898652-88404446	21
0119026783-011093893	31
4298126388-42970436	20
0503882538-050049974	48
6951069615-69518265	54
1996663006-19904959	65
4510514167-45112299	00
9905818693-99009910	54
0669287870-066191684	42
2853792297-28571862	84
4294740646-42902698	97
8193895340-819057939	92
4590941870-45918206	62
1666485814-166523414	41
4384619564-438438822	23
2809141103-280043194	42

#### 23-EKCE-775-RTS

I, the undersigned, certify that a true copy of the attached Order has been served to the following by means of

electronic service on 11/21/2023

ERNEST KUTZLEY, KS ADVOCACY DIRECTOR AARP 6220 SW 29th St. Suite 300 TOPEKA, KS 66614 ekutzley@aarp.org

SHELLY M BASS, SENIOR ATTORNEY ATMOS ENERGY CORPORATION 5430 LBJ FREEWAY 1800 THREE LINCOLN CENTRE DALLAS, TX 75240 shelly.bass@atmosenergy.com

TODD E. LOVE, ATTORNEY CITIZENS' UTILITY RATEPAYER BOARD 1500 SW ARROWHEAD RD TOPEKA, KS 66604 t.love@curb.kansas.gov

SHONDA RABB CITIZENS' UTILITY RATEPAYER BOARD 1500 SW ARROWHEAD RD TOPEKA, KS 66604 s.rabb@curb.kansas.gov

DOROTHY BARNETT CLIMATE & ENERGY PROJECT PO BOX 1858 HUTCHINSON, KS 67504-1858 barnett@climateandenergy.org JAMES G. FLAHERTY, ATTORNEY ANDERSON & BYRD, L.L.P. 216 S HICKORY PO BOX 17 OTTAWA, KS 66067 jflaherty@andersonbyrd.com

JOSEPH R. ASTRAB, ATTORNEY CITIZENS' UTILITY RATEPAYER BOARD 1500 SW ARROWHEAD RD TOPEKA, KS 66604 j.astrab@curb.kansas.gov

DAVID W. NICKEL, CONSUMER COUNSEL CITIZENS' UTILITY RATEPAYER BOARD 1500 SW ARROWHEAD RD TOPEKA, KS 66604 d.nickel@curb.kansas.gov

DELLA SMITH CITIZENS' UTILITY RATEPAYER BOARD 1500 SW ARROWHEAD RD TOPEKA, KS 66604 d.smith@curb.kansas.gov

MELISSA M. BUHRIG, EXEC. VICE PRESIDENT, GEN. COUNSEL & SECRETARY CVR ENERGY, INC. 2277 Plaza Dr., Ste. 500 Sugar Land, TX 77479 mmbuhrig@cvrenergy.com

#### 23-EKCE-775-RTS

JASON T GRAY, ATTORNEY DUNCAN & ALLEN 1730 Rhode Island Ave., NW Suite 700 Washington, DC 20036 jtg@duncanallen.com

DARRIN R. IVES, V.P. REGULATORY AFFAIRS EVERGY METRO, INC D/B/A EVERGY KANSAS METRO One Kansas City Place 1200 Main St., 19th Floor Kansas City, MO 64105 darrin.ives@evergy.com

LESLIE R WINES, SR EXECUTIVE ADMIN ASSISTANT EVERGY METRO, INC D/B/A EVERGY KANSAS METRO One Kansas City Place 1200 Main St., 19th Floor Kansas City, MO 64105 leslie.wines@evergy.com

DANIEL J BULLER, ATTORNEY FOULSTON SIEFKIN LLP 7500 COLLEGE BOULEVARD, STE 1400 OVERLAND PARK, KS 66201-4041 dbuller@foulston.com

LEE M SMITHYMAN, ATTORNEY FOULSTON SIEFKIN LLP 7500 COLLEGE BOULEVARD, STE 1400 OVERLAND PARK, KS 66201-4041 Ismithyman@foulston.com

C. EDWARD WATSON, ATTORNEY FOULSTON SIEFKIN LLP 1551 N. Waterfront Parkway Suite 100 Wichita, KS 67206 cewatson@foulston.com CATHRYN J. DINGES, SR DIRECTOR & REGULATORY AFFAIRS COUNSEL EVERGY KANSAS CENTRAL, INC 818 S KANSAS AVE PO BOX 889 TOPEKA, KS 66601-0889 cathy.dinges@evergy.com

RONALD A. KLOTE, DIRECTOR, REGULATORY AFFAIRS EVERGY METRO, INC D/B/A EVERGY KANSAS METRO ONE KANSAS CITY PLACE 1200 MAIN, 19TH FLOOR KANSAS CITY, MO 64105 ronald.klote@evergy.com

DAVID BANKS, CEM, CEP FLINT HILLS ENERGY CONSULTANT 117 S PARKRIDGE WICHITA, KS 67209 david@fheconsultants.net

SARAH C. OTTO FOULSTON SIEFKIN LLP 7500 COLLEGE BOULEVARD, STE 1400 OVERLAND PARK, KS 662014041 sotto@foulston.com

CONNOR A THOMPSON, ATTORNEY FOULSTON SIEFKIN LLP 7500 COLLEGE BOULEVARD, STE 1400 OVERLAND PARK, KS 66201-4041 cthompson@foulston.com

JAMES P ZAKOURA, ATTORNEY FOULSTON SIEFKIN LLP 7500 COLLEGE BOULEVARD, STE 1400 OVERLAND PARK, KS 66201-4041 jzakoura@foulston.com

#### 23-EKCE-775-RTS

JOHN B. COFFMAN, ATTORNEY AT LAW JOHN B. COFFMAN 871Tuxedo Blvd. St. Louis, MO 63119 john@johncoffman.net ABIGAIL EMERY, PARALEGAL & GRANT SPECIALIST KANSAS CORPORATION COMMISSION 1500 SW ARROWHEAD RD TOPEKA, KS 66604 a.emery@kcc.ks.gov

BRIAN G. FEDOTIN, GENERAL COUNSEL KANSAS CORPORATION COMMISSION 1500 SW ARROWHEAD RD TOPEKA, KS 66604 b.fedotin@kcc.ks.gov

WALKER HENDRIX, LITIGATION COUNSEL KANSAS CORPORATION COMMISSION 1500 SW ARROWHEAD RD TOPEKA, KS 66604 w.hendrix@kcc.ks.gov

AHSAN LATIF, LITIGATION COUNSEL KANSAS CORPORATION COMMISSION 1500 SW ARROWHEAD RD TOPEKA, KS 66604 a.latif@kcc.ks.gov

CARLY MASENTHIN, LITIGATION COUNSEL KANSAS CORPORATION COMMISSION 1500 SW ARROWHEAD RD TOPEKA, KS 66604 c.masenthin@kcc.ks.gov

GLENDA CAFER, ATTORNEY MORRIS LAING EVANS BROCK & KENNEDY 800 SW JACKSON SUITE 1310 TOPEKA, KS 66612-1216 gcafer@morrislaing.com JUSTIN GRADY, CHIEF OF REVENUE REQUIREMENTS, COST OF SERVICE & FINANCE KANSAS CORPORATION COMMISSION 1500 SW ARROWHEAD RD TOPEKA, KS 66604 j.grady@kcc.ks.gov

ANDRIA JACKSON, DEPUTY CHIEF OF REVENUE REQ., COST OF SERVICE & FINANCE KANSAS CORPORATION COMMISSION 1500 SW ARROWHEAD RD TOPEKA, KS 66604 a.jackson@kcc.ks.gov

KRISTINA LUKE-FRY, MANAGING AUDITOR KANSAS CORPORATION COMMISSION 1500 SW ARROWHEAD RD TOPEKA, KS 66604 k.luke-fry@kcc.ks.gov

ROBERT E. VINCENT, MANAGING ATTORNEY KANSAS GAS SERVICE, A DIVISION OF ONE GAS, INC. 7421W. 129TH STREET OVERLAND PARK, KS 66213 robert.vincent@onegas.com

VALERIE SMITH, ADMINISTRATIVE ASSISTANT MORRIS LAING EVANS BROCK & KENNEDY 800 SW JACKSON SUITE 1310 TOPEKA, KS 66612-1216 vsmith@morrislaing.com

#### 23-EKCE-775-RTS

TREVOR WOHLFORD, ATTORNEY MORRIS LAING EVANS BROCK & KENNEDY 800 SW JACKSON SUITE 1310 TOPEKA, KS 66612-1216 twohlford@morrislaing.com

WILL B. WOHLFORD, ATTORNEY MORRIS LAING EVANS BROCK & KENNEDY CHTD 300 N MEAD STE 200 WICHITA, KS 67202-2745 wwohlford@morrislaing.com

TIM OPITZ OPITZ LAW FIRM, LLC 308 E. HIGH STREET SUITE B101 JEFFERSON CITY, MO 65101 tim.opitz@opitzlawfirm.com

FRANK A. CARO, JR., ATTORNEY POLSINELLI PC 900 W 48TH PLACE STE 900 KANSAS CITY, MO 64112 fcaro@polsinelli.com

TIMOTHY J LAUGHLIN, ATTORNEY SCHOONOVER & MORIARTY, LLC 130 N. CHERRY STREET, STE 300 OLATHE, KS 66061 tlaughlin@schoonoverlawfirm.com

ROBERT R. TITUS TITUS LAW FIRM, LLC 7304 W. 130th St. Suite 190 Overland Park, KS 66213 rob@tituslawkc.com RITA LOWE, PARALEGAL MORRIS LAING EVANS BROCK & KENNEDY CHTD 300 N MEAD STE 200 WICHITA, KS 67202-2745 rlowe@morrislaing.com

ASHOK GUPTA, EXPERT NATIONAL RESOURCES DEFENSE COUNCIL 20 N WACKER DRIVE SUITE 1600 CHICAGO, IL 60606 agupta@nrdc.org

PAUL T. DAVIS PAUL DAVIS LAW FIRM, LLC 932 Massachusetts St. Suite 301 Lawrence, KS 66044 pdavis@pauldavislawfirm.com

JARED R. JEVONS, ATTORNEY POLSINELLI PC 900 W 48TH PLACE STE 900 KANSAS CITY, MO 64112 jjevons@polsinelli.com

ROBERT HORTON III TITUS LAW FIRM, LLC 7304 W. 130th St. Suite 190 Overland Park, KS 66213 trey@tituslawkc.com

DEREK S. CASEY TRIPLETT, WOOLF & GARRETSON, LLC 2959 N. Rock Road, Suite 300 Wichita, KS 67226 dscasey@twgfirm.com

#### 23-EKCE-775-RTS

KACEY S MAYES, ATTORNEY TRIPLETT, WOOLF & GARRETSON, LLC 2959 N ROCK RD STE 300 WICHITA, KS 67226 ksmayes@twgfirm.com TIMOTHY E. MCKEE, ATTORNEY TRIPLETT, WOOLF & GARRETSON, LLC 2959 N ROCK RD STE 300 WICHITA, KS 67226 temckee@twgfirm.com

JOHN J. MCNUTT, GENERAL ATTORNEY U.S. ARMY LEGAL SERVICES AGENCY REGULATORY LAW OFFICE 9275 GUNSTON RD., STE. 1300 FORT BELVOIR, VA 22060-5546 john.j.mcnutt.civ@army.mil DAN LAWRENCE, GENERAL COUNSEL - USD 259 UNIFIED SCHOOL DISTRICT 259 903 S EDGEMOOR RM 113 WICHITA, KS 67218 dlawrence@usd259.net

KEVIN K. LACHANCE, CONTRACT LAW ATTORNEY UNITED STATES DEPARTMENT OF DEFENSE ADMIN & CIVIL LAW DIVISION OFFICE OF STAFF JUDGE ADVOCATE FORT RILEY, KS 66442 kevin.k.lachance.civ@army.mil

> /S/ KCC Docket Room KCC Docket Room

# STATE OF MAINE PUBLIC UTILITIES COMMISSION

# DOCKET NO. 2022-00152



# CENTRAL MAINE POWER COMPANY 2022 DISTRIBUTION RATE CASE FILING

RETURN ON EQUITY AND CAPITAL STRUCTURE

August 11, 2022

Testimony and Exhibits of

Ann E. Bulkley

The Brattle Group

On behalf of Central Maine Power Company 83 Edison Drive Augusta, ME 04336

03

# **Table of Contents**

I.	INTRO	DUCTION AND QUALIFICATIONS1
11.	PURPO	DSE AND OVERVIEW OF TESTIMONY1
Ш.	SUMM	ARY OF ANALYSIS AND CONCLUSIONS2
IV.	REGU	LATORY GUIDELINES7
v.	CAPIT	AL MARKET CONDITIONS10
	А.	The Effect of Monetary Policy on Market Dynamics12
	<b>B.</b>	Inflationary Expectations in Current and Projected Market Conditions15
	С.	The Effect of Inflation on Interest Rates and the Investor-Required Return 19
	D.	Expected Performance of Utility Stocks and the Investor-Required ROE
		on Utility Investments23
	Е.	Conclusions
VI.	PROX	Y GROUP SELECTION
VII.	COST	OF EQUITY ESTIMATION
	А.	Importance of Multiple Analytical Approaches
	В.	Discounted Cash Flow (DCF) Model
	C.	Capital Asset Pricing Model42
	D.	Bond Yield Plus Risk Premium Analysis47
VШ	. REGU	LATORY AND BUSINESS RISKS
	А.	Capital Expenditures51
	В.	Regulatory Environment
		1. Cost Recovery Mechanisms
		2. Authorized ROEs
		3. Regulatory Risk
IX.	CAPIT	AL STRUCTURE
x.	MULT	I-YEAR RATE PLAN74
XI.	CONC	LUSIONS AND RECOMMENDATIONs

#### INTRODUCTION AND QUALIFICATIONS 2 Q. Please state your name, affiliation, and business address. 3 My name is Ann E. Bulkley. I am a Principal with The Brattle Group ("Brattle"), located Α. 4 at One Beacon Street, Boston, Massachusetts, 02108. 5 **Q**. On whose behalf are you submitting this testimony? 6 Α. I am submitting this testimony before the Maine Public Utilities Commission 7 ("Commission") on behalf of Central Maine Power Company ("CMP" or the "Company"), which is a wholly-owned subsidiary of Avangrid, Inc. ("Avangrid"), a 8 9 diversified energy and utility company. 10 Please summarize your work experience and educational background. Q. 11 My Curriculum Vitae ("CV") is set forth in Exhibit AEB-1. A. 12 Are you sponsoring any exhibits in support of your testimony? **Q**. 13 Yes. My analyses and recommendations are supported by the data presented in Exhibit Α. 14 AEB-2 through Exhibit AEB-13, which were prepared by me or under my direction. 15 PURPOSE AND OVERVIEW OF TESTIMONY 11. 16 What is the purpose of your testimony? **Q**. 17 Α. The purpose of my testimony is to present evidence and provide a recommendation 18 regarding the appropriate Return on Equity ("ROE")<sup>1</sup> for the Company to be used for

1

Ι.

- 19 ratemaking purposes. My recommendations are reflective of today's record inflation,
- 20 which is expected to persist in the near term and increase the operating risk of the utility,
- 21 as well as the current increasing interest rate environment, which is likewise expected to

<sup>&</sup>lt;sup>+</sup> Throughout this testimony, the terms "ROE" and "cost of equity" are used interchangeably.

1

2

persist in the near term in response to inflation. I also address the appropriateness of the Company's capital structure.

## 3 Q. How is the remainder of your testimony organized?

4 A The remainder of my testimony is organized as follows. Section III provides a summary 5 of my analyses and conclusions. Section IV reviews the regulatory guidelines pertinent to the development of the cost of capital. Section V discusses current and projected 6 7 capital market conditions and the effect of those conditions on CMP's cost of equity. Section VI explains my selection of a proxy group of electric and natural gas utilities. 8 9 Section VII describes my analyses and the analytical basis for the recommendation of the 10 appropriate ROE for CMP. Section VIII provides a discussion of specific regulatory, business, and financial risks that have a direct bearing on the ROE to be authorized for 11 12 the Company in this case. Section IX addresses the Company's capital structure. Section 13 X provides an assessment of the effect of a Multi-Year Rate Plan on the ROE. Section 14 XI presents my conclusions and recommendations for the market cost of equity.

15

### III. SUMMARY OF ANALYSIS AND CONCLUSIONS

- 16 Q. Please provide a brief overview of the analyses that led to your ROE
- 17 recommendation.

18 A. As discussed in more detail in Section VII, I applied the Constant Growth form of the

19 Discounted Cash Flow ("DCF") model, the Capital Asset Pricing Model ("CAPM"), the

- 20 Empirical Capital Asset Pricing Model ("ECAPM"), and the Bond Yield Plus Risk
- 21 Premium analysis. My recommendation also takes into consideration: (1) the Company's
- 22 capital expenditure requirements; and (2) the regulatory environment in which the
- 23 Company operates. While I did not make any specific adjustments to my ROE estimates
- for any of these factors, I did take them into consideration in aggregate when determining

1		where the Company's ROE falls within the range of analytical results. Finally, I
2		considered the Company's projected capital structure as compared to the capital
3		structures of the proxy companies. <sup>2</sup>
4	Q.	Please summarize the key factors considered in your analyses and upon which you
5		base your recommended ROE.
6	Α.	In developing my recommended ROE for CMP, I considered the following:
7		• The <i>Hope</i> and <i>Bluefield</i> decisions <sup>3</sup> that established the standards for determining a
8		fair and reasonable allowed ROE, including consistency of the allowed return with
9		the returns of other businesses having similar risk, adequacy of the return to provide
10		access to capital and support credit quality, and the requirement that the result lead to
11		just and reasonable rates.
12		• The effect of current and projected capital market conditions on investors' return
13		requirements.
14		• The results of several analytical approaches that provide estimates of the Company's
15		cost of equity.
16		• The Company's regulatory, business, and financial risks relative to the proxy group of
17		comparable companies, and the implications of those risks.

 $<sup>^{2}</sup>$  The selection and purpose of developing a group of comparable companies will be discussed in detail in Section VI of my testimony.

<sup>&</sup>lt;sup>3</sup> Federal Power Commission v. Hope Natural Gas Co., 320 U.S. 591 (1944) ("Hope"); Bluefield Waterworks & Improvement Co. v. Public Service Commission of West Virginia, 262 U.S. 679 (1923) ("Bluefield").

# 1 Q. Please explain how you considered these factors.

A. After considering these factors and the results of my analyses, I relied on the range of
results produced by the Constant Growth DCF model, the CAPM and ECAPM, and the
Bond Yield Plus Risk Premium analysis, which are shown in

1	А.	Figure 1. These ROE estimation models produce a wide range of results. My conclusion
2		as to where, within that range of results, CMP's cost of equity falls is based on my
3		assessment of market conditions, and the Company's business and financial risk relative
4		to the proxy group. Although the companies in my proxy group are generally
5		comparable to CMP, each company is unique, and no two companies have the exact same
6		business and financial risk profiles. Accordingly, I considered the Company's business
7		and financial risk in the aggregate in comparison to that of the proxy group companies
8		when determining where the Company's ROE falls within the reasonable range of
9		analytical results to account for any residual differences in risk.
10	Q.	Please summarize the ROE estimation models that you considered to establish the
11		range of ROEs for CMP.

12 A.

- 1 A. Figure 1 summarizes the range of results produced by the Constant Growth DCF, CAPM,
- 2 ECAPM, and Bond Yield Risk Premium analyses.
- 3

# FIGURE 1: SUMMARY OF COST OF EQUITY ANALYTICAL RESULTS



1 As shown in

Figure 1 (and in Exhibit AEB-2), the range of results produced by the ROE estimation models is wide. While it is common to consider multiple models to estimate the cost of equity, it is particularly important when the range of results varies considerably across methodologies. As a result, my ROE recommendation considers the range of results of the Constant Growth DCF model, as well as the results of the CAPM, ECAPM, and Bond Yield Plus Risk Premium analyses. My ROE recommendation also considers CMP's company-specific risk factors and current and prospective capital market conditions.

8

#### Q. Please summarize your conclusions.

9 A. In determining the appropriate ROE and capital structure for a company it is important to
10 consider more than the results of the traditional ROE estimation models. As is discussed
11 in more detail below, it is important to consider the overall market conditions and how
12 those conditions affect the assumptions of the ROE estimation models. In addition, it is
13 necessary to consider the relative risk of the company, in this case, CMP, as compared
14 with the proxy group. The analyses presented in my testimony support the following
15 conclusions:

16 Inflation is expected to persist over the near-term which increases the operating risk 17 of the utility. Additionally, long-term interest rates are expected to increase over the 18 near-term in response to inflation. Utility share prices are inversely related to 19 changes in interest rates. As interest rates rise, it is likely that utility share prices will 20decline. Therefore, the DCF model which relies on current utility share prices is 21 likely understating the cost of equity during the period that CMP's rates will be in 22 effect. This change in market conditions also supports the use of other ROE 23 estimation models such as the CAPM and ECAPM which may be specified using 24 forward looking inputs and thus better reflect expected market conditions.

ROE-8

1	• Equity analysts have noted the increased risk for the utility sector as a result of rising
2	interest rates and therefore expect the sector to underperform over the near-term.
3	• CMP faces relatively greater business and financial risk relative to the proxy group
4	due to the regulatory environment in Maine and the Company's significant capital
5	investments plan.
6	It is reasonable and appropriate to consider all of these factors when estimating, within
7	the model results, the range and estimated investor-required return on equity for CMP.
8	Comparing CMP to the proxy group, it is evident that CMP has a higher overall risk
9	profile than the proxy group, related to the differences in the specific operating risk
10	factors identified in my testimony. Reviewing the analysis summarized in

Figure 1 above, I concluded that a reasonable ROE is within the range of 9.75 to 11.25%. The
 Company is requesting an ROE of 10.20% which below the midpoint of my
 recommended range and is a conservative estimate of the investor-required ROE
 considering the ROE results presented in

1		Figure 1 and the Company's proposed three-year rate plan during which interest rates are
2		expected to increase. As the Company explains elsewhere in testimony, the selection of
3		10.20% is intended to reflect the special economic circumstances affecting its customers
4		at this time.
5	Q.	Please summarize your analysis of the appropriate ratemaking capital structure for
6		the Company.
7	А.	Based on the analysis presented in Section IX of my testimony, I conclude that CMP's
8		proposed 50.00% common equity is reasonable. To determine if CMP's requested capital
9		structure was reasonable, I reviewed the capital structures of the utility subsidiaries of the
10		proxy companies. As shown in Exhibit AEB-13, the results of that analysis demonstrate
11		that the average equity ratios for the utility operating companies of the proxy group range
12		from 44.95% to 63.58%, with an average of 54.10%. Comparing the recommended
13		equity ratio to the proxy group demonstrates that the Company's requested equity ratio is
14		well within the range of equity ratios for the utility operating subsidiaries of the proxy
15		group companies. Further, the Company's proposed equity ratio is reasonable
16		considering the negative effect of the Tax Cuts and Jobs Act of 2017 ("TCJA") and
17		increased capital expenditures on the cash flows and credit metrics of regulated utilities.
18	IV.	REGULATORY GUIDELINES
19	Q.	Please describe the guiding principles to be used in establishing the cost of capital
20		for a regulatory utility.
21	A.	The United States Supreme Court's precedent-setting Hope and Bluefield cases
22		established the standards for determining the reasonableness of a utility's allowed ROE.
23		Among the standards established by the Court in those cases are: (1) consistency with the
24		returns on equity investments in other businesses having similar or comparable risks; (2)

# **ROE-11**

1		adequacy of the return to support credit quality and access to capital; and (3) an
2		understanding that the means of arriving at a fair return are not controlling, only that the
3		end result leads to just and reasonable rates. <sup>4</sup>
4	Q.	Is fixing a fair rate of return just about protecting the utility's interests?
5	A.	No. As the court noted in Bluefield, a proper rate of return not only assures "confidence
6		in the financial soundness of the utility and should be adequate, under efficient and
7		economical management, to maintain and support its credit [but also] enable[s the utility]
8		to raise the money necessary for the proper discharge of its public duties." Bluefield, 262
9		U.S. at 693. As the Court went on to explain in Hope, "[t]he rate-making process
10		involves balancing of the investor and consumer interests." Hope, 320 U.S. at 603.
11	Q.	Has the Commission provided similar guidance in establishing the appropriate return
12		on common equity?
13	А.	Yes, it has. The Commission has consistently recognized the precedent established under
14		Hope and Bluefield. For example, in Docket No. 2019-00092, the Commission stated the
15		following:
16 17 18 19 20		Based on that standard, determining an appropriate ROE for a regulated utility involves determining a market-based cost of equity. For a company that is not publicly traded, such as Northern, the cost of equity is determined to be the return investors expect from alternative investments that present no more and no less risk. In practice, estimating the cost of equity involves developing a comparable group of companies (the so called provy group), for which market
21		המוווטמומטוב צועננט טרעטוווטמוובה דווב הטי-גמווכנו טרטעע צועננט. נטר שוווטר וומואבר

financial risks, and using economic and financial models to set an appropriate
ROE. The Hope-Bluefield standard has long served as the benchmark against
which this Commission measures an appropriate ROE.<sup>5</sup>

<sup>&</sup>lt;sup>4</sup> Hope, 320 U.S. 591; Bluefield, 262 U.S. 679.

<sup>&</sup>lt;sup>5</sup> Northern Utilities, Inc., Request for Approval of Rate Change, Docket No. 2019-00092, Order (Part II) (April 1, 2020) at 51.

1

2

# Q. Why is it important for a utility to be allowed the opportunity to earn an ROE that is adequate to attract capital at reasonable terms?

A. An ROE that is adequate to attract capital at reasonable terms enables the Company to
continue to provide safe, reliable electric service while maintaining its financial integrity.
To the extent the Company is provided the opportunity to earn its market-based cost of
capital, neither customers nor shareholders are disadvantaged.

# Q. Is a utility's ability to attract capital also affected by the ROEs that are authorized for other utilities?

9 A. Yes. Utilities compete directly for capital with other investments of similar risk, which 10 include other natural gas and electric utilities. Therefore, the ROE awarded to a utility sends an important signal to investors regarding whether there is regulatory support for 11 12 financial integrity, dividends, growth, and fair compensation for business and financial 13 risk. The cost of capital represents an opportunity cost to investors. If higher returns are 14 available for other investments of comparable risk, investors have an incentive to direct 15 their capital to those investments. Thus, an authorized ROE that is not in line with authorized ROEs for other natural gas and electric utilities, on a risk-adjusted basis, can 16 17 inhibit the utility's ability to attract capital for investment in Maine.

18 Q. What are your conclusions regarding regulatory guidelines?

A. The ratemaking process is premised on the principle that a utility must have the
opportunity to recover the return of, and the market-required return on, its invested
capital. Because utility operations are capital-intensive, regulatory decisions should
enable the utility to attract capital at reasonable terms under a variety of economic and
financial market conditions; doing so balances the long-term interests of the utility and its
customers.

**ROE-13** 

1		The financial community carefully monitors the current and expected financial
2		condition of utility companies and the regulatory framework in which they operate. In
3		that respect, the regulatory framework is one of the most important factors in both debt
4		and equity investors' assessments of risk. The Commission's order in this proceeding,
5		therefore, should establish rates that provide the Company with the opportunity to earn an
6		ROE that is: (1) adequate to attract capital at reasonable terms under a variety of
7		economic and financial market conditions; (2) sufficient to ensure good financial
8		management and firm integrity; and (3) commensurate with returns on investments in
9		enterprises with similar risk. To the extent CMP is authorized the opportunity to earn its
10		market-based cost of capital, the proper balance is achieved between customers' and
11		shareholders' interests.
12	V.	CAPITAL MARKET CONDITIONS
13	Q.	Why is it important to consider capital market conditions in the estimation of the
13 14	Q.	Why is it important to consider capital market conditions in the estimation of the investor-required return on equity?
13 14 15	<b>Q.</b> A.	Why is it important to consider capital market conditions in the estimation of the investor-required return on equity? The ROE estimation models rely on market data that are either specific to the proxy
13 14 15 16	<b>Q.</b> A.	Why is it important to consider capital market conditions in the estimation of theinvestor-required return on equity?The ROE estimation models rely on market data that are either specific to the proxygroup, in the case of the DCF model, or to the expectations of market risk, in the case of
13 14 15 16 17	<b>Q.</b> A.	Why is it important to consider capital market conditions in the estimation of theinvestor-required return on equity?The ROE estimation models rely on market data that are either specific to the proxygroup, in the case of the DCF model, or to the expectations of market risk, in the case ofthe CAPM. Prevailing market conditions at the time the analysis is performed affect the
13 14 15 16 17 18	<b>Q.</b>	Why is it important to consider capital market conditions in the estimation of the investor-required return on equity? The ROE estimation models rely on market data that are either specific to the proxy group, in the case of the DCF model, or to the expectations of market risk, in the case of the CAPM. Prevailing market conditions at the time the analysis is performed affect the results of the ROE estimation models. While the ROE that is established in a rate
13 14 15 16 17 18 19	<b>Q.</b>	Why is it important to consider capital market conditions in the estimation of the investor-required return on equity? The ROE estimation models rely on market data that are either specific to the proxy group, in the case of the DCF model, or to the expectations of market risk, in the case of the CAPM. Prevailing market conditions at the time the analysis is performed affect the results of the ROE estimation models. While the ROE that is established in a rate proceeding is intended to be forward-looking, current and projected market data,
<ol> <li>13</li> <li>14</li> <li>15</li> <li>16</li> <li>17</li> <li>18</li> <li>19</li> <li>20</li> </ol>	<b>Q.</b>	Why is it important to consider capital market conditions in the estimation of theinvestor-required return on equity?The ROE estimation models rely on market data that are either specific to the proxygroup, in the case of the DCF model, or to the expectations of market risk, in the case ofthe CAPM. Prevailing market conditions at the time the analysis is performed affect theresults of the ROE estimation models. While the ROE that is established in a rateproceeding is intended to be forward-looking, current and projected market data,specifically stock prices, dividends, growth rates and interest rates are used in the ROE
<ol> <li>13</li> <li>14</li> <li>15</li> <li>16</li> <li>17</li> <li>18</li> <li>19</li> <li>20</li> <li>21</li> </ol>	<b>Q.</b>	Why is it important to consider capital market conditions in the estimation of the investor-required return on equity? The ROE estimation models rely on market data that are either specific to the proxy group, in the case of the DCF model, or to the expectations of market risk, in the case of the CAPM. Prevailing market conditions at the time the analysis is performed affect the results of the ROE estimation models. While the ROE that is established in a rate proceeding is intended to be forward-looking, current and projected market data, specifically stock prices, dividends, growth rates and interest rates are used in the ROE estimation models to estimate the required return for the subject company.
<ol> <li>13</li> <li>14</li> <li>15</li> <li>16</li> <li>17</li> <li>18</li> <li>19</li> <li>20</li> <li>21</li> <li>22</li> </ol>	<b>Q.</b>	Why is it important to consider capital market conditions in the estimation of the investor-required return on equity? The ROE estimation models rely on market data that are either specific to the proxy group, in the case of the DCF model, or to the expectations of market risk, in the case of the CAPM. Prevailing market conditions at the time the analysis is performed affect the results of the ROE estimation models. While the ROE that is established in a rate proceeding is intended to be forward-looking, current and projected market data, specifically stock prices, dividends, growth rates and interest rates are used in the ROE estimation models to estimate the required return for the subject company. As is discussed in the remainder of this section, because current market conditions
<ol> <li>13</li> <li>14</li> <li>15</li> <li>16</li> <li>17</li> <li>18</li> <li>19</li> <li>20</li> <li>21</li> <li>22</li> <li>23</li> </ol>	Q.	Why is it important to consider capital market conditions in the estimation of the investor-required return on equity? The ROE estimation models rely on market data that are either specific to the proxy group, in the case of the DCF model, or to the expectations of market risk, in the case of the CAPM. Prevailing market conditions at the time the analysis is performed affect the results of the ROE estimation models. While the ROE that is established in a rate proceeding is intended to be forward-looking, current and projected market data, specifically stock prices, dividends, growth rates and interest rates are used in the ROE estimation models to estimate the required return for the subject company. As is discussed in the remainder of this section, because current market conditions

**ROE-14** 

1 range and recommended ROE for a future period. If investors do not expect current 2 market conditions to be sustained in the future, it is possible that the ROE estimation 3 models will not provide an accurate estimate of investors' required return during that rate 4 period. Therefore, it is important to consider projected market data to estimate the return 5 for that forward-looking period. 6 **Q**. What factors are affecting the cost of equity for regulated utilities in the current and 7 prospective capital markets? 8 The cost of equity for regulated utility companies is being affected by several factors in Α. 9 the current and prospective capital markets, including: (1) persistently high inflation, 10 (2) changes in monetary policy, and (3) rising interest rates. These factors affect the assumptions used in the ROE estimation models. In this section, I discuss each of these 11 12 factors and how they affect the models used to estimate the cost of equity for regulated 13 utilities. 14 Q. What effect do current and prospective market conditions have on the cost of equity 15 for the Company? 16 Α. As is discussed in more detail in the remainder of this section, the combination of 17 persistently high inflation, the Federal Reserve's changes in monetary policy, and the dramatic shifts in market conditions resulting from political influences all contribute to an 18 19 expectation of increased market risk and an increase in the cost of the investor-required 20 return on equity. It is essential that these factors be considered in setting a forward-21 looking cost of equity. Inflation is currently at its highest level seen in approximately 40 22 years. Interest rates, which have increased from the pandemic lows seen in 2020 are 23 expected to continue to increase in direct response to the Federal Reserve's monetary 24 policy. Since there is a strong historical inverse correlation between interest rates and the

#### **ROE-15**

1419

1		share prices of utility stocks (share prices of utility stocks typically fall when interest
2		rates rise), it is reasonable to expect that investors' required ROE for utility companies
3		will also continue to increase. Therefore, ROE estimates based solely on current market
4		conditions will understate the ROE required by investors during the future period that the
5		Company's rates determined in this proceeding will be in effect.
6		A. The Effect of Monetary Policy on Market Dynamics
7	Q.	Please summarize the monetary policy actions of the Federal Reserve in response to
8		the economic effects of COVID-19.
9	A.	In response to the COVID-19 pandemic, the Federal Reserve:
10		• Decreased the Federal Funds rate twice in March 2020, resulting in a target range of
11		0.00% to 0.25%;
12		• Increased its holdings of both Treasury and mortgaged-back securities;
13		• Started expansive programs to support credit to large employers—the Primary Market
14		Corporate Credit Facility to provide liquidity for new issuances of corporate bonds;
15		and the Secondary Market Corporate Credit Facility to provide liquidity for
16		outstanding corporate debt issuances; and
17		• Supported the flow of credit to consumers and businesses through the Term Asset-
18		Backed Securities Loan Facility.
19		In addition, Congress also passed the Coronavirus Aid, Relief, and Economic Security
20		("CARES") Act in March 2020, the Consolidated Appropriations Act, 2021 in December
21		2020, and the American Rescue Plan Act in March 2021, which included \$2.2 trillion,
22		\$900 billion, and \$1.9 trillion, respectively, in fiscal stimulus aimed at also mitigating the
23		economic effects of COVID-19. These expansive monetary and fiscal programs

# **ROE-16**

1 2 mitigated the economic effects of the COVID-19 pandemic and provided additional support as the economy recovers from the COVID-19 recession.

# Q. How did the accommodative monetary and fiscal policy affect the U.S. economy? A. The expansive monetary and fiscal policy programs resulted in a strong economic

5 recovery in 2021 from the COVID-19 induced recessionary period in 2020. In fact, 6 according to the Bureau of Economic Analysis, real GDP grew by 5.7% in 2021 driven primarily by a 7.9% increase in personal consumption expenditures.<sup>6</sup> Moreover, the 7 unemployment rate decreased from a high of 14.7% in April 2020 to 3.9% as of 8 December 2021.<sup>7</sup> Finally, as I will discuss in more detail below, the economic recovery 9 10 has also included a substantial increase in inflation with the year-over-year ("YOY") change in the Consumer Price Index ("CPF") at 9.0% in June 2022. As noted below, 11 12 there are several factors that have contributed to the currently high inflation including 13 supply chain disruptions related to the COVID-19 pandemic and energy price increases 14 related to the Ukraine war. The strong economic recovery along with the increase in 15 inflation resulted in the Federal Reserve normalizing monetary policy and beginning to 16 remove the accommodative policy programs that it used to mitigate the effect of COVID-19. 17

# 18 Q. Is the Federal Reserve normalizing monetary policy?

- 19 A. The dramatic increase in inflation has prompted the Federal Reserve to pursue an
- 20 aggressive normalization of monetary policy, removing the accommodating policy

<sup>&</sup>lt;sup>6</sup> Bureau of Economic Analysis, News Release, February 24, 2022, at 8.

<sup>&</sup>lt;sup>7</sup> Bureau of Labor Statistics. https://data.bls.gov/timeseries/LNS14000000.

1	programs used to mitigate the effects of COVID-19. As of the June 15, 2022 meeting,
2	the Federal Reserve has taken the following actions:
3	• Completed its taper of Treasury bond and mortgage-backed securities purchase; <sup>8</sup>
4	• Increased the target federal funds rate from $0.00 - 0.25\%$ to $0.25 - 0.50\%$ at the
5	March 16, 2022 meeting, <sup>9</sup> from $0.25 - 0.50\%$ to 0.75 to 1.00% at the May 4, 2022
6	meeting, $^{10}$ and then from 0.75 to 1.00% to 1.50% to 1.75% at the June 15, 2022
7	meeting; <sup>11</sup>
8	• Forecasted a total of seven additional 25 basis point rate increases in 2022 and two 25
9	basis point rate increases in 2023 which resulted a median forecast of the federal
10	funds rate of 3.4% and 3.8%, respectively; <sup>12</sup> and
11	• Began reducing its holdings of Treasury and mortgage-backed securities on June 1,
12	2022. <sup>13</sup> The Federal Reserve is reducing the size of its balance sheet by only
13	reinvesting principal payments on owned securities after the total amount of
14	payments received exceeds a defined cap. For Treasury Securities, the cap is set at
15	\$30 billion per month for the first three months and \$60 billion per month after the
16	first three months. The cap for mortgage-backed securities is set at \$17.5 billion per
17	month for the first three months and \$35 billion per month thereafter. <sup>14</sup>

<sup>&</sup>lt;sup>8</sup> Federal Reserve Bank of New York, https://www.newyorkfed.org/markets/domestic-market-operations/monetarypolicy-implementation/treasury-securities/treasury-securities-operational-details#monthly-details.

<sup>&</sup>lt;sup>9</sup> Federal Reserve, Press Release, March 16, 2022.

<sup>&</sup>lt;sup>10</sup> Federal Reserve, Press Release, May 4, 2022.

<sup>&</sup>lt;sup>11</sup> Federal Reserve, Press Release, June 15, 2022.

<sup>&</sup>lt;sup>12</sup> Federal Reserve, Summary of Economic Projections, June 15, 2022, at 2.

<sup>&</sup>lt;sup>13</sup> Federal Reserve, Press Release, May 4, 2022.

<sup>&</sup>lt;sup>14</sup> Federal Reserve, Plans for Reducing the Size of the Federal Reserve's Balance Sheet, Press Release, May 4, 2022.

1	Q.	Has the Federal Reserve provided additional support for the expectation that it will
2		continue to aggressively normalize monetary policy to reduce inflation?
3	Α.	Yes. Specifically, Federal Reserve Chairman Powell noted at his press conference on
4		June 15, 2022 that reducing inflation to the long-term goal of 2% was the primary
5		objective and that additional rate increases will be necessary with a 50 or 75 basis point
6		increase likely needed at the next meeting:
7 8 9 10 11 12 13 14 15 16 17		Over coming months, we will be looking for compelling evidence that inflation is moving down, consistent with inflation returning to 2 percent. We anticipate that ongoing rate increases will be appropriate; the pace of those changes will continue to depend on the incoming data and the evolving outlook for the economy. Clearly, today's 75 basis point increase is an unusually large one, and I do not expect moves of this size to be common. From the perspective of today, either a 50 or 75 basis point increase seems most likely at our next meeting. We will, however, make our decisions meeting by meeting, and we will continue to communicate our thinking as clearly as we can. Our overarching focus is using our tools to bring inflation back down to our 2 percent goal and to keep longer-term inflation expectations well anchored.
18 19 20 21 22 23 24 25 26		Making appropriate monetary policy in this uncertain environment requires a recognition that the economy often evolves in unexpected ways. Inflation has obviously surprised to the upside over the past year, and further surprises could be in store. We therefore will need to be nimble in responding to incoming data and the evolving outlook. And we will strive to avoid adding uncertainty in what is already an extraordinarily challenging and uncertain time. We are highly attentive to inflation risks and determined to take the measures necessary to restore price stability. The American economy is very strong and well positioned to handle tighter monetary policy. <sup>15</sup>
27		<b>B.</b> Inflationary Expectations in Current and Projected Market Conditions
28	Q.	Is the increase in inflation significant?

29 A. Yes. As shown in Figure 2, the YOY change in the CPI published by the Bureau of

30 Labor statistics was 1.37% in January 2021. However, since that time, and particularly

31 since the start of 2022, inflation has increased steadily, reaching a high of 9.0% YOY

<sup>&</sup>lt;sup>15</sup> Federal Reserve, Transcript of Chair Powell's Press Conference Opening Statement, June 15, 2022, at 4-5.

change in June 2022, which is the largest 12-month increase since 1981 and significantly
 greater than any level seen since January 2008. The 9.0% YOY change in the CPI in
 June 2022 is down only slightly from the high in March 2022.



## FIGURE 2: CONSUMER PRICE INDEX—YOY PERCENT CHANGE JANUARY 2008–JUNE 2022<sup>16</sup>

### 4 Q. What are the expectations for inflation over the near-term?

A. In his press conference following the June 15, 2022, meeting, Chairman Powell noted
that "[w]e at the Fed understand the hardship high inflation is causing. We are strongly
committed to bringing inflation back down, and we are moving expeditiously to do so".<sup>17</sup>
Therefore, investors expect inflation to remain elevated over the near-term. One measure
of investors' expectations regarding inflation is the breakeven inflation rate calculated as
the spread between the yield on a Treasury bond and the yield on a Treasury InflationProtected bond, since a Treasury Inflation-Protected bond would account for the effect of

<sup>&</sup>lt;sup>16</sup> Bureau of Labor Statistics, shaded area indicates a recession.

<sup>&</sup>lt;sup>17</sup> Federal Reserve, Transcript of Chair Powell's Press Conference, (June 15, 2022), at 1.

inflation. The maturity of the bond selected would then reflect investors' views of
 inflation during the holding period of the bond.

3 For example, the 5-year breakeven inflation rate calculated as the spread between 4 the 5-year Treasury bond yield and the 5-year Treasury Inflation-Protected bond yield 5 would reflect investors' expectations of inflation over the next 5 years. As shown in Figure 3 below, the 5-year breakeven inflation rate is currently greater than any level 6 seen since January 2003. Furthermore, the 5-year breakeven inflation rate as of May 31, 7 8 2022 was 2.96% indicating that investors expect inflation will remain well above the 9 Federal Reserve's 2% target over the next 5 years. Therefore, inflation will remain 10 elevated for the duration of the Company's proposed multi-year rate plan.



FIGURE 3: 5-YEAR BREAKEVEN INFLATION RATE JANUARY 2003–MAY 2022<sup>18</sup>

—5-γear Breakeven Inflation Rate

There are many factors as to why inflation is expected to remain elevated. *Kiplinger* recently noted a few factors including supply shortages due to COVID-19 and Russia's
 war in Ukraine which led them to forecast an inflation rate of 8% for 2022.
 Gasoline prices continued their strong rise in June, and the overall inflation
 rate is likely to stay at the same high level in June. It should peak at about 9%

rate is likely to stay at the same high level in June. It should peak at about 9% by the end of the summer, then decline gradually after that, ending the year at 6 7 about 8.0% before dropping to 3-4% next year. The higher cost of housing 8 will still keep inflation rates elevated for some time to come. Gasoline prices 9 and heating costs are likely to stay high for a good while because of the war in 10 Ukraine, but energy prices are likely to peak during the summer and ease after that. The price of cars and trucks will also stay at a high level until the 11 12 semiconductor shortage ends sometime next year. Continued spot shortages of various items will drive their prices up, adding to the overall inflation rate. 13 The latest is a shortage of tampons.<sup>19</sup> 14

<sup>&</sup>lt;sup>18</sup> Federal Reserve Bank of St. Louis, 5-Year Breakeven Inflation Rate [T5YIE], retrieved from FRED, Federal Reserve Bank of St. Louis; https://fred.stlouisfed.org/series/T10YIE, May 31, 2022.

<sup>&</sup>lt;sup>19</sup> Payne, David, "Inflation Should Peak This Summer at About 9%," *Kiplinger*, June 10, 2022.

1

### C. The Effect of Inflation on Interest Rates and the Investor-Required Return

2

# Q. What effect will inflation have on long-term interest rates?

A. Inflation and the Federal Reserve's normalization of monetary policy will likely result in
 increases in long-term interest rates. Specifically, inflation reduces the purchasing power
 of the future interest payments an investor expects to receive over the duration of the
 bond. This risk increases the longer the duration of the bond. As a result, if investors
 expect increased levels of inflation, they will require higher yields to compensate for the
 increased risk of inflation, which means interest rates will increase.

9

10

# Q. Have the yields on long-term government bonds increased in response to inflation and the Federal Reserve's normalization of monetary policy?

11 A. Yes, they have. As discussed above, at the December 2021, January 2022, March 2022, 12 May 2022 and June 2022 meetings, the Federal Reserve noted its continued concerns over the sustained increased levels of inflation. In addition, starting at the December 13 2021 meeting and continuing through the June 2022 meeting, the Federal Reserve 14 15 accelerated the process of normalizing monetary policy to respond to inflation. As shown 16 in Figure 4, since the Federal Reserve's December 2021 meeting, the yield on 10-year 17 Treasury bond has doubled, increasing from 1.47% on December 15, 2021 to 2.85% on May 31, 2022. The increase is due to the Federal Reserve's announcements at the 18 19 December 2021, January 2022, March 2022 and May 2022 meetings as well as investors' 20 expectations regarding the Federal Reserve's announcement at the June 2022 meeting, 21 and the continued increased levels of inflation that are now expected to persist much longer than the Federal Reserve and investors had originally projected. 22



#### FIGURE 4: 10-YEAR TREASURY BOND YIELD JANUARY 2021–MAY 2022<sup>20</sup>

### 1 Q. What have equity analysts said about long-term government bond yields?

A. Leading equity analysts have noted that they expect the yields on long-term government
bonds to remain elevated through at least the end of 2022. According to views of equity
analysts summarized in Figure 5, the yield on the 10-year Treasury Bond is expected to
range from 3.15% to 4.00% by the end of 2022, which is 26 to 111 basis points greater
than the current 30-day average yield on the 10-year Treasury Bond as of May 31, 2022
of 2.89%. Furthermore, as of June 14, 2022, the yield on the 10-year Treasury was
trading at 3.49%.

<sup>20</sup> S&P Capital IQ Pro.
	10-year U.S. Treasury Yield		
Bank	30-day Average as of May 31, 2022	2022 Forecast	
Advocate Capital Management <sup>21</sup>	2.89%	4.00%	
Goldman Sachs <sup>22</sup>	2.89%	3,30%	
Blue Chip Financial Forecasts (Consensus Estimate) <sup>23</sup>	2.89%	3,20%	
BMO Economics <sup>24</sup>	2.89%	3.15%	

#### FIGURE 5: EQUITY ANALYSTS FORECAST OF THE 10-YEAR TREASURY YIELD

## Q. Have you considered any additional indicators that may imply long-term interest rates are expected to increase?

3	А.	Yes. In addition to the yields on Treasury bonds, I considered the net position of
4		commercials (i.e., banks) in U.S. Treasury Bond futures contracts as reported in the
5		Commitment of Traders ("COT") Report produced by the Commodity Futures Trading
6		Commission ("CFTC"). A net position is defined as the total number of long positions in
7		a futures contract minus the total number of short positions in a futures contract. A long
8		position means that an investor agrees to purchase an asset in the future at a specified
9		price today and therefore profits if the price of the underlying asset increases.
10		Conversely, a short position is when an investor agrees to sell an asset at a time in the
11		future at a specified price today and profits if the price of the asset declines. Therefore, if

<sup>&</sup>lt;sup>21</sup> MarketWatch, "This bond expert who called the spike in U.S. yields forecasts the 10-year to reach 4%," May 7, 2022. https://www.marketwatch.com/story/this-bond-expert-who-called-the-spike-in-u-s-yields-forecasts-the-10-year-to-reach-4-11651843223.

<sup>&</sup>lt;sup>22</sup> Pollard, Amelia. "Goldman Lifts Yield Forecasts, Sees 10-Year Treasuries at 3.3%." Bloomberg.com, May 12, 2022.

<sup>&</sup>lt;sup>23</sup> Blue Chip Financial Forecasts, Vol. 41, No. 6, June 1, 2022, at 2.

<sup>&</sup>lt;sup>24</sup> BMO Economics, "Rates Scenario for May 11, 2022," May 11, 2022.

banks are increasing the number of short positions and thus have a declining net position, 2 the banks are assuming that the price of the asset will decline. As shown in Figure 6, the 3 net position of banks in U.S. Treasury Bonds has been decreasing since the end of 2020. 4 Therefore, banks are forecasting a decrease in the price of long-term government bonds 5 and thus the yields (which are inversely related to the price) to increase over the near-6 term.

1

#### FIGURE 6: COMMITMENT OF TRADERS REPORT—NET POSITION OF COMMERCIALS (I.E., BANKS) IN U.S. TREASURY BOND FUTURES CONTRACTS<sup>25</sup>



<sup>&</sup>lt;sup>25</sup> Commitment of Traders Report, as of May 31, 2022, https://www.cftc.gov/MarketReports/CommitmentsofTraders/HistoricalCompressed/index.htm

1		D. Expected Performance of Utility Stocks and the Investor-Required ROE on
2		Utility Investments
3	Q.	Are utility share prices correlated to changes in the yields on long-term government
4		bonds?
5	Α.	Yes. Interest rates and utility share prices are inversely correlated which means, for
6		example, that an increase in interest rates will result in a decline in the share prices of
7		utilities. For example, Goldman Sachs and Deutsche Bank recently examined the
8		sensitivity of share prices of different industries to changes in interest rates over the past
9		five years. Both Goldman Sachs and Deutsche Bank found that utilities had one of the
10		strongest negative relationships with bond yields (i.e., increases in bond yields resulted in
11		the decline of utility share prices). <sup>26</sup>
12	Q.	Have electric utility stock prices recently increased?
13	A.	Yes. Utility stock prices had trended down as interest rates moved higher; however, as a
14		result of the political turmoil associated with the war in Ukraine, investors have recently
15		returned to utility stocks as a safe haven seeking to lower risk, resulting in higher electric
16		utility stock prices and thus lower dividend yields. <sup>27</sup>
17	Q.	How do equity analysts expect the utilities sector to perform in an increasing
18		interest rate environment?
19	А.	Even with the recent increase in electric utility stock prices, equity analysts project that

utilities will underperform the broader market as interest rates increase. For example, in

<sup>&</sup>lt;sup>26</sup> Lee, Justina. "Wall Street Is Rethinking the Treasury Threat to Big Tech Stocks." Bloomberg.com, 11 Mar. 2021, www.bloomberg.com/news/articles/2021-03-11/wall-street-is-rethinking-the-treasury-threat-to-big-tech-stocks.

<sup>&</sup>lt;sup>27</sup> Sonenshine, Jacob. "Utilities Have Been Soaring as Treasuries Get Crushed. That Isn't Supposed to Happen." Barrons.com, April 11, 2022, https://www.barrons.com/articles/utilities-treasury-yields-outlook-51649457572?mod=hp\_INTERESTS\_bonds&refsec=hp\_INTERESTS\_bonds

1		its most recent Big Money Poll, which closed in mid-April 2022 and surveyed 112
2		money managers regarding the outlook for the next twelve months, the professional
3		investors surveyed by Barron's selected the utility sector as the least attractive of all
4		industries for investment. <sup>28</sup> In addition, Fidelity recently recommended underweighting
5		the utility sector and noted that it classified the sector as underweight due to a
6		combination of "poor fundamentals and expensive valuations." <sup>29</sup> Furthermore, regarding
7		the recent increase in utility share prices, Fidelity stated that:
8 9 10 11 12		Energy stocks have garnered a lot of attention, but in February utilities was the only sector with monthly returns in the 90th percentile of its historical range. In the past, powerful utilities rallies have signaled investors getting too defensive. The market typically has gained, and utilities have underperformed, in 12-month periods after top-decile monthly relative returns for the sector. <sup>30</sup>
13	Q.	Have you reviewed any market indicators that may imply that utilities will
13 14	Q.	Have you reviewed any market indicators that may imply that utilities will underperform over the near-term?
13 14 15	<b>Q.</b> A.	Have you reviewed any market indicators that may imply that utilities will underperform over the near-term? Yes, I have. As discussed above, the utility sector is considered a "bond proxy" or a
13 14 15 16	<b>Q.</b>	Have you reviewed any market indicators that may imply that utilities willunderperform over the near-term?Yes, I have. As discussed above, the utility sector is considered a "bond proxy" or asector that investors view as a "safe haven" alternative to bonds, and changes in utility
13 14 15 16 17	<b>Q.</b> A.	Have you reviewed any market indicators that may imply that utilities willunderperform over the near-term?Yes, I have. As discussed above, the utility sector is considered a "bond proxy" or asector that investors view as a "safe haven" alternative to bonds, and changes in utilitystock prices are therefore inversely related to changes in interest rates. For example, the
13 14 15 16 17 18	<b>Q.</b>	Have you reviewed any market indicators that may imply that utilities willunderperform over the near-term?Yes, I have. As discussed above, the utility sector is considered a "bond proxy" or asector that investors view as a "safe haven" alternative to bonds, and changes in utilitystock prices are therefore inversely related to changes in interest rates. For example, theutility sector tends to perform well when interest rates are low since the dividend yields
13 14 15 16 17 18 19	<b>Q.</b>	Have you reviewed any market indicators that may imply that utilities willunderperform over the near-term?Yes, I have. As discussed above, the utility sector is considered a "bond proxy" or asector that investors view as a "safe haven" alternative to bonds, and changes in utilitystock prices are therefore inversely related to changes in interest rates. For example, theutility sector tends to perform well when interest rates are low since the dividend yieldsfor utilities offer investors the prospect of higher returns when compared to the yields on
<ol> <li>13</li> <li>14</li> <li>15</li> <li>16</li> <li>17</li> <li>18</li> <li>19</li> <li>20</li> </ol>	<b>Q.</b>	Have you reviewed any market indicators that may imply that utilities willunderperform over the near-term?Yes, I have. As discussed above, the utility sector is considered a "bond proxy" or asector that investors view as a "safe haven" alternative to bonds, and changes in utilitystock prices are therefore inversely related to changes in interest rates. For example, theutility sector tends to perform well when interest rates are low since the dividend yieldsfor utilities offer investors the prospect of higher returns when compared to the yields onlong-term government bonds. Conversely, the utility sector underperforms as the yields
<ol> <li>13</li> <li>14</li> <li>15</li> <li>16</li> <li>17</li> <li>18</li> <li>19</li> <li>20</li> <li>21</li> </ol>	<b>Q.</b>	Have you reviewed any market indicators that may imply that utilities will underperform over the near-term? Yes, I have. As discussed above, the utility sector is considered a "bond proxy" or a sector that investors view as a "safe haven" alternative to bonds, and changes in utility stock prices are therefore inversely related to changes in interest rates. For example, the utility sector tends to perform well when interest rates are low since the dividend yields for utilities offer investors the prospect of higher returns when compared to the yields on long-term government bonds. Conversely, the utility sector underperforms as the yields

<sup>&</sup>lt;sup>28</sup> Jasinski, Nicholas. "Bearish Now, Bullish Later: How Investors Are Sizing up Stocks," Barron's updated April 24, 2022.

<sup>&</sup>lt;sup>29</sup> Fidelity, "Top sectors to watch in Q2," May 4, 2022.

<sup>&</sup>lt;sup>30</sup> Ibid.

examined the difference ("yield spread") between the dividend yields of utility stocks and
the yields on long-term government bonds from January 2010 through May 2022. 1
selected the dividend yield on the S&P Utilities Index as the measure of the dividend
yields for the utility sector and the yield on the 10-year Treasury Bond as the estimate of
the yield on long-term government bonds.

As shown in Figure 7, the yield spread as of May 31, 2022, was 0.00% indicating 6 7 that the yield on the 10-year Treasury Bond is equivalent to the dividend yield for the 8 S&P Utilities Index. Furthermore, the current yield spread of 0.00% is well below the 9 long-term average since January 2010 of 1.45%. Given that the yield spread is currently 10 well below the long-term average as well as the expectation that interest rates will continue to increase, it is reasonable to conclude that utility sector will most likely 11 12 underperform over the near-term. This is because investors that purchased utility stocks 13 as an alternative to the lower yields on long-term government bonds would otherwise be 14 inclined to rotate back into government bonds, particularly as the yields on long-term 15 government bonds continue to increase, thus resulting in a decrease in the share prices of utilities. 16

#### FIGURE 7: YIELD SPREAD BETWEEN THE DIVIDEND YIELD ON THE S&P UTILITIES INDEX AND THE YIELD ON THE 10-YEAR TREASURY BOND – JANUARY 2010 – MAY 2022<sup>31</sup>



# Q. What is the significance of the inverse relationship between interest rates and utility share prices in the current market?

A. As discussed above, the Federal Reserve is currently normalizing monetary policy in
response to inflation which actions are expected to increase long-term government bond
yields. If interest rates increase as expected, then the share prices of utilities will decline.
If the prices of utility stocks decline, then the DCF model, which relies on historical
averages of share prices, is likely to understate the cost of equity. For example, Figure 8,
below summarizes the effect of price on the dividend yield in the Constant Growth DCF
model.

<sup>&</sup>lt;sup>31</sup> Bloomberg Professional and S&P Capital IQ Pro.

#### FIGURE 8: THE EFFECT OF A DECLINE IN STOCK PRICES ON THE CONSTANT GROWTH DCF MODEL



A decline in stock prices will increase the dividend yields and thus the estimate of the 1 2 ROE produced by the Constant Growth DCF model. Therefore, this expected change in 3 market conditions supports consideration of the range of ROE results produced by the 4 mean to mean-high DCF results since the mean DCF results would likely understate the 5 cost of equity during the period that the Company's rates will be in effect. Moreover, 6 prospective market conditions warrant consideration of other ROE estimation models 7 such as the CAPM and ECAPM, which may better reflect expected market conditions. 8 For example, two out of three inputs to the CAPM (*i.e.*, the market risk premium and 9 risk-free rate) are forward-looking.

10 E. Conclusions

Q. What are your conclusions regarding the effect of current market conditions on the
cost of equity for the Company?

A. Over the near-term, investors expect long-term interest rates to increase in response to
 continued elevated levels of inflation and the Federal Reserve's normalization of

- 15 monetary policy. Because the share prices of utilities are inversely correlated to interest
- 16 rates, an increase in long-term government bond yields will likely result in a decline in
- 17 utility share prices, which is the reason a number of equity analysts expect the utility
- 18 sector to underperform over the near-term. The expected underperformance of utilities

**ROE-31** 

1		means that DCF models using recent historical data likely underestimate investors'
2		required return over the period that rates will be in effect. This change in market
3		conditions also supports the use of other ROE estimation models such as the CAPM and
4		the ECAPM, which may better reflect expected market conditions.
5	VI.	PROXY GROUP SELECTION
6	Q.	Why have you used a group of proxy companies to estimate the Cost of Equity for
7		CMP?
8	Α.	In this proceeding, we focus on estimating the cost of equity for an electric utility
9		company that is not itself publicly traded. Because the cost of equity is a market-based
10		concept and because CMP's operations do not make up the entirety of a publicly traded
11		entity, it is necessary to establish a group of companies that are both publicly traded and
12		comparable to CMP in certain fundamental business and financial respects to serve as its
13		"proxy" in the ROE estimation process.
14		Even if CMP were a publicly-traded entity, it is possible that transitory events
15		could bias its market value over a given period. A significant benefit of using a proxy
16		group is that it moderates the effects of unusual events that may be associated with any
17		one company. The proxy companies used in my analyses all possess a set of operating
18		and risk characteristics that are substantially comparable to the Company, and thus
19		provide a reasonable basis to derive and estimate the appropriate ROE for CMP.
20	Q.	Please provide a brief profile of CMP.

A. CMP is an electric transmission and distribution utility, wholly owned by Avangrid,
serving approximately 646,000 customers in central and southern Maine covering close

1		to 11,000 square miles. <sup>32</sup> The company's service territory encompasses most of Maine's
2		largest cities, including Portland, Lewiston, Brunswick and Augusta. CMP's
3		transmission and distribution system represents approximately 21.37% of the total rate
4		base of Avangrid's Networks subsidiary. <sup>33</sup> In 2021, the Company delivered
5		approximately 9,297,000 MWh of electricity <sup>34</sup> and had total distribution operating
6		revenues of \$313 million. <sup>35</sup> Additionally, CMP had a distribution rate base in 2021 of
7		\$1,014 million <sup>36</sup> which consisted of 21,954 miles of overhead lines, and 1,780 miles of
8		underground lines. <sup>37</sup> CMP currently has an investment grade long-term rating of A
9		(Outlook: Stable) from S&P <sup>38</sup> , and A2 (Outlook: Stable) from Moody's. <sup>39</sup>
10	Q.	How did you select the companies included in your proxy group?
11	Α.	In prior cases (i.e., Docket No. 2015-00360 and Docket No. 2013-00443), the
12		Commission Staff ("Staff") noted that including companies in the proxy group that own
13		natural gas distribution operations or using a separate proxy group comprised of natural
14		gas distribution companies is appropriate for the purposes of comparing to an electric
15		utility that does not own any generation. <sup>40</sup> Specifically, Staff stated in Docket No. 2015-
16		00360 that "[1]ike distribution and transmission of electricity through poles and wires,

<sup>&</sup>lt;sup>32</sup> CMP website.

<sup>36</sup> Avangrid, Inc., 2021 Form 10-K, at 9.

<sup>33</sup> Avangrid, Inc., 2021 Form 10-K, at 8.

<sup>&</sup>lt;sup>34</sup> Ibid.

<sup>35</sup> Company provided data.

<sup>&</sup>lt;sup>37</sup> Id. at 18.

<sup>&</sup>lt;sup>38</sup> S&P Capital IQ Pro, accessed June 30, 2022

<sup>&</sup>lt;sup>39</sup> Moody's Investors Service, accessed June 30, 2022.

<sup>&</sup>lt;sup>40</sup> Emera Maine, Request for Approval of a Proposed Rate Increase, Docket No. 2015-00360, Bench Analysis at 6 (June 2, 2016); Bangor Hydro Electric Company and Maine Public Service Company, Proposed Increase in Distribution Rates, Docket No. 2013-00443, Bench Analysis, at 7 (March 17, 2014).

1	transportation of gas through pipes presents a similar risk profile to electric T&D
2	utilities."41 Moreover, in Docket No. 2018-00194 for CMP, Staff developed a proxy
3	group that included natural gas distribution companies for the purposes of estimating the
4	cost of equity for CMP. <sup>42</sup> In recognition of Staff's position and in order to minimize the
5	issues that are disputed in this case, I considered for inclusion in the proxy group,
6	companies that were classified by Value Line as Natural Gas Distribution Companies.
7	Therefore, to develop my proxy group, I began with the companies that Value Line
8	classifies as Electric Utilities and Natural Gas Distribution Companies. That combined
9	group includes 46 domestic U.S. utilities. I simultaneously applied the following
10	screening criteria to establish a risk-comparable proxy group that includes both electric
11	utility companies and natural gas distribution companies that:
12	• Pay consistent quarterly cash dividends, because companies that do not cannot be
13	analyzed using the Constant Growth DCF model;
14	• Have investment grade long-term issuer ratings from S&P and/or Moody's;
15	• Are covered by at least two utility industry analysts;
16	• Have positive long-term earnings growth forecasts from at least two utility
17	industry equity analysts;
18	• Derive more than 70.00% of their total operating income from regulated
19	operations; and

<sup>&</sup>lt;sup>41</sup> Emera Maine, Request for Approval of a Proposed Rate Increase, Docket No. 2015-00360, Bench Analysis, at 6-7 (June 2, 2016).

<sup>&</sup>lt;sup>42</sup> Central Maine Power Company, Investigation into the Rates and Revenue Requirements of Central Maine Power Company, Docket No. 2018-00194, Bench Analysis, at 42 (February 22, 2019).

1		• Were not parties to a merger or transformative transaction during the relevant
2		analytical periods.
3	Q.	Did you include Avangrid in your analysis?
4	A.	No. It is my practice to exclude the subject company, or its parent holding company,
5		from the proxy group to avoid circular logic that otherwise would occur.
6	Q.	Did you exclude other companies from the proxy group?
7	A.	Yes. I also excluded Pinnacle West Capital Corporation ("PNW") and Hawaiian Electric
8		Industries, Inc. ("HE") from the proxy group. For PNW, the share price decreased
9		approximately 24% over a two-month period from October through November 2021
10		resulting from a negative regulatory decision for its largest operating company, Arizona
11		Public Service Company. Therefore, similar to the reason that I exclude transformative
12		transactions; because the stock price can be affected by one-time events, I also excluded
13		PNW from the proxy group.
14		HE's operations are concentrated on the islands of Hawaii; therefore, the
15		company faces geographic concentration risk. As HE noted in the company's 2020
16		Form10-K:
17		The Company is subject to the risks associated with the geographic
18		concentration of its businesses and current lack of interconnections that could
10		result in service interruptions at the Utilities or higher default rates on loans
20		held by ASB [American Savings Bank]. <sup>43</sup>
21		The increased risk of service interruptions resulting from HE's geographic location which
22		could result in revenue loss and increased costs is a risk unique to HE and would not
23		apply to utilities located on the U.S. mainland. Furthermore, HE's unregulated
24		operations which represent approximately 33% of the company's operation income in

<sup>&</sup>lt;sup>43</sup> Hawaii Electric Industries, Inc., 2021 Form 10-K, at 23.

7	Q.	What is the composition of your proxy group?
6		my proxy group considering HE's unique geographical risks.
5		company's geographic concentration in Hawaii.45 As a result, I have excluded HE from
4		adverse economic or political event, ASB could face severe financial effects given the
3		consumer and commercial loans are to customers in Hawaii. If Hawaii were to face an
2		Bank ("ASB").44 ASB also only operates on Hawaii; thus, all of the company's
1		2021 are concentrated in the banking sector through the ownership of American Savings

A. The screening criteria discussed above is shown in Exhibit AEB-3 and resulted in a proxy
group consisting of the companies shown in Figure 9 below.

Company	Ticker
ALLETE, Inc.	ALE
Alliant Energy Corporation	LNT
Ameren Corporation	AEE
American Electric Power Company, Inc.	AEP
Atmos Energy Corporation	ATO
Avista Corporation	AVA
Black Hills Corporation	BKH
CMS Energy Corporation	CMS
Consolidated Edison, Inc.	ED
Duke Energy Corporation	DUK
Edison International	EIX
Entergy Corporation	ETR
Evergy, Inc.	EVRG
Eversource Energy	ES
IDACORP, Inc.	IDA
MGE Energy, Inc.	MGEE
NextEra Energy, Inc.	NEE
NiSource Inc.	NI
Northwest Natural Gas Company	NWN

FIGURE 9: PROXY GROUP

<sup>44</sup> Id. at 86,

<sup>&</sup>lt;sup>45</sup> *Id.* at 20.

Company	Ticker
NorthWestern Corporation	NWE
OGE Energy Corporation	OGE
ONE Gas, Inc.	OGS
Otter Tail Corporation	OTTR
Portland General Electric Company	POR
Public Service Enterprise Group Inc.	PEG
Southern Company	SO
Spire, Inc.	SR
Wisconsin Energy Corporation	WEC
Xcel Energy Inc.	XEL

#### 1 VII. **COST OF EQUITY ESTIMATION**

#### 2 О. Please briefly discuss the ROE in the context of the regulated rate of return.

3 A. The rate of return ("ROR") for a regulated utility is based on its weighted average cost of capital, in which the costs of the individual sources of capital are weighted by their 4 5 respective percentages of total capitalization of the utility. The ROE included in the ROR is weighted by the percentage of common equity in the regulated utility's 6 7 ratemaking capital structure.

8 О.

17

#### How is the required ROE determined?

9 While the cost of debt can be directly observed, the cost of equity and the required ROE A. 10 are market-based and, therefore, must be estimated based on observable market 11 information. The required ROE is determined by using one or more analytical techniques 12 that rely on market data to quantify investor expectations regarding the range of required 13 equity returns. Informed judgment is applied, based on the results of those analyses, to 14 determine where within the range of results the cost of equity for a company falls. As a 15 general proposition, the key consideration in determining the cost of equity is to ensure 16 that the methodologies employed reasonably reflect investors' views of the financial

markets, the proxy group companies, and the subject company's risk profile.

**Q**.

#### What methods did you use to determine the Company's ROE?

A. I considered the results of the Constant Growth DCF model, the CAPM, the ECAPM,
and the Bond Yield Plus Risk Premium Analysis. As discussed in more detail below, a
reasonable ROE estimate appropriately considers alternative methodologies and the
reasonableness of their individual and collective results.

6

#### A. Importance of Multiple Analytical Approaches

#### 7 Q. Why is it important to use more than one analytical approach?

8 Because the cost of equity is not directly observable, it must be estimated based on both Α. 9 quantitative and qualitative information. When faced with the task of estimating the cost 10 of equity, analysts and investors are inclined to gather and evaluate as much relevant data as reasonably can be analyzed. As a result, a number of models have been developed to 11 12 estimate the cost of equity. For that reason, I use multiple approaches to estimate the cost 13 of equity. As a practical matter, however, all of the models available for estimating the 14 cost of equity are subject to limiting assumptions or other methodological constraints. 15 Consequently, many finance texts recommend using multiple approaches when estimating the cost of equity. For example, Copeland, Koller, and Murrin<sup>46</sup> suggest using 16 the CAPM and Arbitrage Pricing Theory model, while Brigham and Gapenski<sup>47</sup> 17 recommend the CAPM, DCF, and "bond yield plus risk premium" approaches. 18 19 Q. Do current market conditions justify using more than one analytical approach? 20Α. Yes. Interest rates have increased and are expected to continue to increase from the lows 21 as a result of the COVID-19 pandemic. Given the inverse relationship between interest

<sup>&</sup>lt;sup>46</sup> Tom Copeland, Tim Koller and Jack Murrin, Valuation: Measuring and Managing the Value of Companies, 3rd Ed. (New York: McKinsey & Company, Inc., 2000), at 214.

<sup>&</sup>lt;sup>47</sup> Eugene Brigham, Louis Gapenski, Financial Management: Theory and Practice, 7th Ed. (Orlando: Dryden Press, 1994), at 341.

1		rates and utility share prices, the dividend yields of utilities are expected to increase over
2		the near-term. Therefore, the current low dividend yields for utilities result in DCF cost
3		of equity estimates that are understating the forward-looking cost of equity. The CAPM
4		and Bond Yield Plus Risk Premium method offer some balance to the sensitivity of the
5		DCF model to Treasury yields. Low interest rates might also affect the CAPM in two
6		ways: (1) the risk-free rate is lower, and (2) because the market risk premium is a
7		function of interest rates, (i.e., it is the return on the broad stock market less the risk-free
8		interest rate), the risk premium should move higher when interest rates are lower.
9		However, when applied appropriately, the CAPM will take into account the relationship
10		between ROE and interest rates through the market risk premium component. Therefore,
11		it is important to use multiple analytical approaches to ensure that the ROE results reflect
12		the market conditions that are expected during the period that CMP's rates will be in
13		effect. Given the expectation that interest rates will increase, it is important to moderate
14		the impact that the current lower interest rates are having on the ROE estimates,
15		especially the DCF analysis, and where possible consider using projected market data in
16		the models to estimate the return for the forward-looking period.
17	Q.	Are you aware of any regulatory commissions that have recognized the importance
18		of considering the results of multiple models?
19	Α.	Yes, several regulatory commissions consider the results of multiple ROE estimation
20		methodologies such as the DCF, CAPM, and ECAPM in determining the authorized
21		ROE, including the Federal Energy Regulatory Commission ("FERC"),48 Minnesota

<sup>&</sup>lt;sup>48</sup> FERC Opinion No. 569-A, Order on Rehearing, May 21, 2020, para. 112, 140.

1	Public Utilities Commission ("Minnesota PUC"),49 the Michigan Public Service
2	Commission ("Michigan PSC"), <sup>50</sup> the Iowa Utilities Board ("IUB"), <sup>51</sup> the Washington
3	Utilities and Transportation Commission ("Washington UTC"),52 and the New Jersey
4	Board of Public Utilities ("NJBPU"). <sup>53</sup> For example, FERC issued Opinion No. 569-A,
5	Order on Rehearing, on May 21, 2020, in which FERC reviewed prior decisions and
6	found that investors rely on the DCF, CAPM, and risk premium approaches to make
7	investment decisions. <sup>54</sup> Therefore the FERC concluded that the results of each of these
8	models should be given equal weight, as the "evidence does not indicate that there is a
9	clearly superior model for estimating cost of equity that should be given more weight
10	than the others."55
11	Additionally, the Washington UTC has repeatedly emphasized that it "places
12	value on each of the methodologies used to calculate the cost of equity and does not find
13	it appropriate to select a single method as being the most accurate or instructive."56 The

14 Washington UTC has also explained that "[f]inancial circumstances are constantly

<sup>&</sup>lt;sup>49</sup> Docket No. G011/GR-17-563, Findings of Fact, Conclusions and Order, at 27; Docket No. E015/GR-16-664, Findings of Fact, Conclusions and Order, at 60-61.

<sup>&</sup>lt;sup>50</sup> Michigan Public Service Commission Order, DTE Gas Company, Case No. U-18999, September 13, 2018, at 45-47.

<sup>&</sup>lt;sup>51</sup> Iowa Utilities Board, Iowa-American Water Company, RPU-2016-0002, Final Decision and Order issued February 27, 2017, at 35.

<sup>&</sup>lt;sup>52</sup> Wash, Utils. & Transp. Comm'n v. PacifiCorp, Docket UE-130043, Order 05, n. 89 (Dec. 4, 2013); Wash. Utils. & Transp. Comm'n v. PacifiCorp, Docket UE-100749, Order 06, ¶ 91 (March 25, 2011).

<sup>&</sup>lt;sup>53</sup> NJBPU Docket No. ER12111052, OAL Docket No. PUC16310-12, Order Adopting Initial Decision with Modifications and Clarifications, March 18, 2015, at 71.

<sup>&</sup>lt;sup>54</sup> FERC Opinion No. 569-A, Order on Rehearing, May 21, 2020, para. 112, 140.

<sup>55</sup> Id. at para, 141.

<sup>&</sup>lt;sup>56</sup> Wash. Utils. & Transp. Comm'n v. PacifiCorp, Docket UE-130043, Order 05, n. 89 (Dec. 4, 2013).

shifting and changing, and we welcome a robust and diverse record of evidence based on a variety of analytics and cost of capital methodologies."<sup>57</sup>

- Finally, in its recent order for DTE Gas Company ("DTE Gas") in Case No.
  U-18999, the Michigan PSC considered the results of each of the models presented by the
  ROE witnesses, which included the DCF, CAPM, and ECAPM in the determination of
  the authorized ROE.<sup>58</sup> The Commission also considered authorized ROEs in other states,
  increased volatility in capital markets and the company-specific business risks of DTE
  Gas.
- 9 **O**.

#### Q. What are your conclusions about the results of the DCF and CAPM models?

10 Α. Recent market data that is used as the basis for the assumptions for both models have been affected by market conditions. As a result, relying exclusively on historical 11 12 assumptions in these models, without considering whether these assumptions are consistent with investors' future expectations, will underestimate the cost of equity that 13 14 investors would require over the period that the rates in this case are to be in effect. In 15 this instance, relying on the historically low dividend yields that are not expected to 16 continue over the period that the new rates will be in effect will underestimate the ROE for CMP. 17

Furthermore, as discussed in Section V above, long-term interest rates have increased since August 2020 and this trend is expected to continue as the Federal Reserve normalizes monetary policy in response to increased inflation. Therefore, the use of current averages of Treasury bond yields as the estimate of the risk-free rate in the

<sup>&</sup>lt;sup>57</sup> Wash. Utils. & Transp. Comm'n v. PacifiCorp. Docket UE-100749, Order 06, ¶91 (March 25, 2011).

<sup>&</sup>lt;sup>58</sup> Michigan Public Service Commission Order, DTE Gas Company, Case No. U-18999, September 13, 2018, at 45-47.

1 CAPM is not appropriate since recent market conditions are not expected to continue 2 over the long-term. Instead, analysts should rely on projected yields of Treasury Bonds 3 in the CAPM. The projected Treasury Bond yields result in CAPM estimates that are 4 more reflective of the market conditions that investors expect during the period that the 5 Company's rates will be in effect.

6

11

15

#### B. Discounted Cash Flow (DCF) Model

7 Q. Please describe the DCF approach.

8 A. The DCF approach is based on the theory that a stock's current market price represents
9 the present value of all expected future cash flows. In its most general form, the DCF
10 model is expressed as follows:

$$P_0 = \frac{D_1}{(1+k)} + \frac{D_2}{(1+k)^2} + \dots + \frac{D_{\infty}}{(1+k)^{\infty}}$$
[1]

Where P<sub>0</sub> represents the current market stock price, D<sub>1</sub> ... D<sub>n</sub> are all expected future
dividends, and k is the discount rate, or required ROE. Equation [1] is a standard present
value calculation that can be simplified and rearranged into the following form:

$$k = \frac{D_0(1+g)}{P_0} + g$$
 [2]

Equation [2] is often referred to as the Constant Growth DCF model in which the first term is the expected dividend yield and the second term is the expected long-term growth rate.

#### 19 Q. What assumptions are required for the Constant Growth DCF model?

20 A. The Constant Growth DCF model requires the following assumptions: (1) a constant

- 21 growth rate for earnings and dividends; (2) a stable dividend payout ratio; (3) a constant
- 22 price-to-earnings ("P/E") ratio; and (4) a discount rate greater than the expected growth

1		rate. To the extent any of these assumptions is violated, considered judgment and/or
2		specific adjustments should be applied to the results.
3	Q.	What market data did you use to calculate the dividend yield in your Constant
4		Growth DCF model?
5	Α.	The dividend yield in my Constant Growth DCF model is based on the proxy companies'
6		current annual dividends and average closing stock prices over the 30-, 90-, and 180-
7		trading days as of May 31, 2022.
8	Q.	Why did you use three averaging periods for stock prices?
9	A.	In my Constant Growth DCF model, I use an average of recent trading days to calculate
10		the price term $(P_0)$ in the DCF model to ensure that the ROE is not skewed by anomalous
11		events that may affect stock prices on any given trading day. The averaging period
12		should also be reasonably representative of expected capital market conditions over the
13		long-term. However, as discussed above, recent market data is not representative of
14		expected market conditions over the long-term. Therefore, the results of my Constant
15		Growth DCF model using historical data may underestimate the forward-looking cost of
16		equity. As a result, I place more weight on the median to median-high results produced
17		by my Constant Growth DCF model.
18	Q.	Did you make any adjustments to the dividend yield to account for periodic growth
19		in dividends?
20	Α.	Yes, I did. Because utility companies tend to increase their quarterly dividends at
21		different times throughout the year, it is reasonable to assume that dividend increases will
22		be evenly distributed over calendar quarters. Given that assumption, it is reasonable to
23		apply one-half of the expected annual dividend growth rate for purposes of calculating
24		the expected dividend yield component of the DCF model. This adjustment ensures that

**ROE-43** 

1447

1		the expected first-year dividend yield is, on average, representative of the coming twelve-
2		month period, and does not overstate the aggregated dividends to be paid during that
3		time.
4	Q.	Why is it important to select appropriate measures of long-term growth in applying
5		the DCF model?
6	Α.	In its Constant Growth form, the DCF model (i.e., Equation [2]) assumes a single growth
7		estimate in perpetuity. To reduce the long-term growth rate to a single measure, one
8		must assume that the payout ratio remains constant and that earnings per share, dividends
9		per share, and book value per share all grow at the same constant rate. Over the long run,
10		however, dividend growth can only be sustained by earnings growth. Therefore, it is
11		important to incorporate a variety of sources of long-term earnings growth rates into the

12 Constant Growth DCF model.

## Q. What sources of long-term growth rates did you rely on in your Constant Growth DCF model?

A. My Constant Growth DCF model incorporates three sources of long-term earnings
growth rates: (1) Zacks Investment Research; (2) Thomson First Call (provided by
Yahoo! Finance); and (3) Value Line Investment Survey.

18 Q. How did you calculate the range of results for the Constant Growth DCF model?

A. I calculated the low result for my DCF model using the minimum growth rate (*i.e.*, the
 lowest of the First Call, Zacks, and Value Line earnings growth rates) for each of the
 proxy group companies. Thus, the low result reflects the minimum DCF result for the
 proxy group. I used a similar approach to calculate the high results, using the highest
 growth rate for each proxy group company. The mean results were calculated using the
 average growth rates from all sources.

1 Q. Please summarize the results of your Constant Growth DCF analyses.

2	Α.	Figure 10 (see also Exhibit AEB-4) presents the range of results produced by my proxy
3		group. As shown in Figure 10, for the proxy group, the median and mean DCF results
4		range from 9.07% to 9.35%, and the median high and mean high results are in the range
5		of 9.80% to 10.47%. While I also summarize the median low and mean low DCF results,
6		given the expected underperformance of utility stocks and thus the likelihood that the
7		DCF model is understating the cost of equity, I do not believe it is appropriate to consider
8		the low DCF results at this time.

### FIGURE 10: CONSTANT GROWTH DISCOUNTED CASH FLOW RESULTS<sup>59</sup>

	Constant Growt	h DCF	
	Mean Low	Mean	Mean High
30-Day Average	8.06%	9.17%	10.28%
90-Day Average	8.12%	9.24%	10.35%
180-Day Average	8.24%	9.35%	10.47%
	Median Low	Median	Median High
30-Day Average	7.94%	9.07%	9.80%
90-Day Average	8.03%	9.11%	9.91%
180-Day Average	8.30%	9.21%	10.06%

#### 9 Q. What are your conclusions about the results of the Constant Growth DCF model?

10 A. As discussed previously, one primary assumption of the DCF model is a constant P/E 11 ratio. That assumption is heavily influenced by the market price of utility stocks. Since 12 utility stocks are expected to underperform the broader market over the near-term as 13 interest rates increase, it is important to consider the results of the DCF models with 14 caution. This means that the results of the DCF models, which rely on historical stock

<sup>&</sup>lt;sup>59</sup> See Exhibit AEB-4.

1		prices, are below where they would be expected to be going forward during the period in
2		which the rates for the Company will be in effect. Therefore, while I have given weight
3		to the results of the Constant Growth DCF model, my recommendation also gives weight
4		to the results of other ROE estimation models.
5		C. Capital Asset Pricing Model
6	Q.	Please briefly describe the Capital Asset Pricing Model ("CAPM")
7	A.	The CAPM is a risk premium approach that estimates the cost of equity for a given
8		security as a function of a risk-free return plus a risk premium to compensate investors
9		for the non-diversifiable or "systematic" risk of that security. This second component is
10		the product of the market risk premium and the Beta coefficient, which measures the
11		relative riskiness of the security being evaluated.
12		The CAPM is defined by four components, each of which must theoretically be a
13		forward-looking estimate:
14		$k_e = r_f + \beta(r_m - r_f) $ [3]
15		where:
16		$k_e$ = the required market ROE
17		$\beta$ = Beta coefficient of an individual security
18		$r_{f}$ = the risk-free rate of return
19		$r_m$ = the required return on the market as a whole
20		In this specification, the term $(r_m - r_f)$ represents the market risk premium. According to
21		the theory underlying the CAPM, investors should be concerned only with systematic or
22		non-diversifiable risk because unsystematic risk can be diversified away. Non-
23		diversifiable risk is measured by the Beta coefficient, which is defined as:

$$\beta = \frac{Covariance(r_e, r_m)}{Variance(r_m)}$$
[4]

1		The variance of the market return, noted in Equation [4], is a measure of the uncertainty
2		of the general market, and the covariance between the return on a specific security and
3		the market reflects the extent to which the return on that security will respond to a given
4		change in the market return.
5	Q.	What risk-free rate did you use in your CAPM analysis?
6	А.	I used three estimates of the yield on Treasury bonds: (1) the current 30-day average
7		yield on 30-year Treasury bonds (3.02%);60 (2) the projected 30-year Treasury yield for
8		Q3 2022 through Q3 2023 (3.48%); <sup>61</sup> and (3) the projected 30-year Treasury yield for the
9		period 2024-2028 (3.80%). <sup>62</sup>
10	Q.	Would you place more weight on one of these scenarios?
11	Α.	Yes. Based on current market conditions, I place more weight on the results of the
11 12	Α.	Yes. Based on current market conditions, I place more weight on the results of the projected yields on the 30-year Treasury bonds. As discussed previously, the estimation
11 12 13	A.	Yes. Based on current market conditions, I place more weight on the results of the projected yields on the 30-year Treasury bonds. As discussed previously, the estimation of the cost of equity in this case should be forward-looking because it is the return that
11 12 13 14	A.	Yes. Based on current market conditions, I place more weight on the results of the projected yields on the 30-year Treasury bonds. As discussed previously, the estimation of the cost of equity in this case should be forward-looking because it is the return that investors would receive over the future rate period. Therefore, the inputs and
<ol> <li>11</li> <li>12</li> <li>13</li> <li>14</li> <li>15</li> </ol>	A.	<ul> <li>Yes. Based on current market conditions, I place more weight on the results of the</li> <li>projected yields on the 30-year Treasury bonds. As discussed previously, the estimation</li> <li>of the cost of equity in this case should be forward-looking because it is the return that</li> <li>investors would receive over the future rate period. Therefore, the inputs and</li> <li>assumptions used in the CAPM analysis should reflect the expectations of the market at</li> </ul>
<ol> <li>11</li> <li>12</li> <li>13</li> <li>14</li> <li>15</li> <li>16</li> </ol>	Α.	Yes. Based on current market conditions, I place more weight on the results of the projected yields on the 30-year Treasury bonds. As discussed previously, the estimation of the cost of equity in this case should be forward-looking because it is the return that investors would receive over the future rate period. Therefore, the inputs and assumptions used in the CAPM analysis should reflect the expectations of the market at that time. While I have included the results of a CAPM analysis that relies on the current
<ol> <li>11</li> <li>12</li> <li>13</li> <li>14</li> <li>15</li> <li>16</li> <li>17</li> </ol>	A.	Yes. Based on current market conditions, I place more weight on the results of the projected yields on the 30-year Treasury bonds. As discussed previously, the estimation of the cost of equity in this case should be forward-looking because it is the return that investors would receive over the future rate period. Therefore, the inputs and assumptions used in the CAPM analysis should reflect the expectations of the market at that time. While I have included the results of a CAPM analysis that relies on the current average risk-free rate, this analysis fails to take into consideration the effect of the

<sup>&</sup>lt;sup>60</sup> Bloomberg Professional.

<sup>&</sup>lt;sup>61</sup> Blue Chip Financial Forecasts, Vol. 41, No. 6, June 1, 2022, at 2.

<sup>&</sup>lt;sup>62</sup> Blue Chip Financial Forecasts, Vol. 41, No. 6, June 1, 2021, at 14.

**Q**.

#### What beta coefficients did you use in your CAPM analysis?

A. As shown in Exhibit AEB-5, I used the Beta coefficients for the proxy group companies
as reported by Bloomberg and Value Line. The Beta coefficients reported by Bloomberg
were calculated using ten years of weekly returns relative to the S&P 500 Index. Value
Line's calculation is based on five years of weekly returns relative to the New York
Stock Exchange Composite Index.

Additionally, as shown in Exhibit AEB-6, I also considered an additional CAPM
analysis which relies on the long-term average utility Beta coefficient for the companies
in my proxy group. The long-term average utility Beta coefficient was calculated as an
average of the Value Line Beta coefficients for the companies in my proxy group from
2013 through 2021.

#### 12 Q. How did you estimate the Market Risk Premium in the CAPM?

13 I estimated the Market Risk Premium ("MRP") as the difference between the implied Α. 14 expected equity market return and the risk-free rate. As shown in Exhibit AEB-7, the 15 expected return on the S&P 500 Index is calculated using the Constant Growth DCF 16 model for the companies in the S&P 500 Index. In my calculation of the market return, I 17 included companies in the S&P 500 that: (1) had either a dividend yield or Value Line 18 long-term earnings projection; and (2) had a Value Line long-term earnings growth rate 19 that was greater than 0% and less than or equal to 20%. Based on an estimated market 20 capitalization-weighted dividend yield of 1.65% and a weighted long-term growth rate of 21 11.11%, the estimated required market return for the S&P 500 Index is 12.86%.







7

Q. Did you consider another form of the CAPM in your analysis?

8 A. Yes. I have also considered the results of an Empirical CAPM ("ECAPM" or

### 9 alternatively referred to as the Zero-Beta CAPM)<sup>64</sup> in estimating the cost of equity for

<sup>&</sup>lt;sup>63</sup> Depicts total annual returns on large company stocks, as reported in the 2022 Duff & Phelps SBBI Yearbook.

<sup>&</sup>lt;sup>64</sup> See e.g., Roger A. Morin, New Regulatory Finance, Public Utilities Reports, Inc., 2006, at 189.

1	CMP. The ECAPM calculates the product of the adjusted Beta coefficient and the
2	market risk premium and applies a weight of 75.00% to that result. The model then
3	applies a 25.00% weight to the market risk premium, without any effect from the Beta
4	coefficient. The results of the two calculations are summed, along with the risk-free rate,
5	to produce the ECAPM result, as noted in Equation [5] below:
	$k_{\rm c} = r_{\rm f} + 0.75\beta(r_{\rm m} - r_{\rm f}) + 0.25(r_{\rm m} - r_{\rm f}) $ [5]
6	where:
7	$k_{e}$ = the required market ROE
8	$\beta$ = Adjusted Beta coefficient of an individual security
9	$r_f$ = the risk-free rate of return
10	$r_m$ = the required return on the market as a whole
11	In essence, the Empirical form of the CAPM addresses the tendency of the "traditional"
12	CAPM to underestimate the cost of equity for companies with low Beta coefficients such
13	as regulated utilities. In that regard, the ECAPM is not redundant to the use of adjusted
14	Betas; rather, it recognizes the results of academic research indicating that the risk-return
15	relationship is different (in essence, flatter) than estimated by the CAPM, and that the
16	CAPM underestimates the "alpha," or the constant return term. <sup>65</sup>
17	As with the CAPM, my application of the ECAPM uses the forward-looking
18	market risk premium estimates, the three yields on 30-year Treasury securities noted
19	earlier as the risk-free rate, and the Bloomberg, Value Line and long-term average Beta
20	coefficients.

1 **Q**.

#### What are the results of your CAPM analyses?

- 2 A. As shown in Figure 12 (see also Exhibit AEB-5), my traditional CAPM analysis produces
- 3 a range of returns from 10.20% to 11.61%. The ECAPM analysis results range from
- 4 10.87% to 11.92%.

	Current Risk-Free Rate (3.02%)	Q3 2022 – Q3 2023 Projected Risk-Free Rate (3.48%)	2024-2028 Projected Risk- Free Rate (3.80%)
	CAPM		
Value Line Beta	11,50%	11,57%	11.61%
Bloomberg Beta	10.79%	10.89%	10.95%
Long-Term Avg. Beta	10.20%	10.33%	10.41%
	ECAPM		
Value Line Beta	11.84%	11.89%	11.92%
Bloomberg Beta	11.31%	11.38%	11.43%
Long-Term Avg. Beta	10.87%	10.96%	11.03%

#### FIGURE 12: CAPM RESULTS

5

#### D. Bond Yield Plus Risk Premium Analysis

#### 6 Q. Please describe the Bond Yield Plus Risk Premium approach.

7 Α. In general terms, this approach is based on the fundamental principle that equity investors 8 bear the residual risk associated with equity ownership and therefore require a premium 9 over the return they would have earned as a bondholder. That is, because returns to 10 equity holders have greater risk than returns to bondholders, equity investors must be 11 compensated to bear that risk. Risk premium approaches, therefore, estimate the cost of 12 equity as the sum of the equity risk premium and the yield on a particular class of bonds. 13 In my analysis, I used actual authorized returns for electric utility companies as the 14 historical measure of the cost of equity to determine the risk premium. 15 Q. Are there other considerations that should be addressed in conducting this analysis? 16 Yes. It is important to recognize both academic literature and market evidence indicating Α.

17 that the equity risk premium (as used in this approach) is inversely related to the level of

**ROE-51** 

1	interest rates. That is, as interest rates increase (decrease), the equity risk premium
2	decreases (increases). Consequently, it is important to develop an analysis that: (1)
3	reflects the inverse relationship between interest rates and the equity risk premium; and
4	(2) relies on recent and expected market conditions. Such an analysis can be developed
5	based on a regression of the risk premium as a function of U.S. Treasury bond yields. If
6	we let authorized ROEs for electric utilities serve as the measure of required equity
7	returns and define the yield on the long-term U.S. Treasury bond as the relevant measure
8	of interest rates, the risk premium simply would be the difference between those two
9	points. <sup>66</sup>

#### 10 Q. Is the Bond Yield Plus Risk Premium analysis relevant to investors?

A. Yes. Investors are aware of ROE awards in other jurisdictions, and they consider those
 awards as a benchmark for a reasonable level of equity returns for utilities of comparable
 risk operating in other jurisdictions. Because my Bond Yield Plus Risk Premium analysis

14 is based on authorized ROEs for utility companies relative to corresponding Treasury

15 yields, it provides relevant information to assess the return expectations of investors.

#### 16 Q. What did your Bond Yield Plus Risk Premium analysis reveal?

17 A. As shown in Figure 13 below, from 1992 through May 2022, there was a strong negative

18 relationship between risk premia and interest rates. To estimate that relationship, I

#### 19 conducted a regression analysis using the following equation:

$$RP = a + b(T) \tag{6}$$

<sup>&</sup>lt;sup>66</sup> See e.g., S. Keith Berry, "Interest Rate Risk and Utility Risk Premia during 1982–93," *Managerial and Decision Economics*, Vol. 19, No. 2 (March, 1998), in which the author used a methodology similar to the regression approach described below, including using allowed ROEs as the relevant data source, and came to similar conclusions regarding the inverse relationship between risk premia and interest rates. *See also* Robert S. Harris, "Using Analysts' Growth Forecasts to Estimate Shareholders Required Rates of Return," *Financial Management*, Spring 1986, at 66.

1	Where
2	RP = Risk Premium (difference between allowed ROEs and the yield on 30-year
3	U.S. Treasury bonds)
4	a = intercept term
5	b = slope term
6	T = 30-year U.S. Treasury bond yield
7	Data regarding allowed ROEs were derived from 885 electric utility rate cases from 1992
8	through May 2022 as reported by Regulatory Research Associates ("RRA").67 This
9	equation's coefficients were statistically significant at the 99.00% level.



FIGURE 13: RISK PREMIUM RESULTS

10 As shown on Exhibit AEB-8, based on the current 30-day average of the 30-year U.S.

11 Treasury bond yield (*i.e.*, 3.02%), the risk premium would be 6.81%, resulting in an

<sup>&</sup>lt;sup>67</sup> This analysis began with a total of 1,383 cases and was screened to eliminate limited issue rider cases, transmission-only cases, and cases that were silent with respect to the authorized ROE. After applying those screening criteria, the analysis was based on data for 885 cases.

1		estimated ROE of 9.83%. Based on the near-term (Q3 2022-Q3 2023) projections of the
2		30-year U.S. Treasury bond yield (i.e., 3.48%), the risk premium would be 6.65%,
3		resulting in an estimated ROE of 10.04%. Based on longer-term (2024-2028) projections
4		of the 30-year U.S. Treasury bond yield (i.e., 3.80%), the risk premium would be 6.38%,
5		resulting in an estimated ROE of 10.18%.
6	Q.	How did the results of the Bond Yield Plus Risk Premium inform your
7		recommended ROE for CMP?
8	Α.	I have considered the results of the Bond Yield Plus Risk Premium analysis in setting my
9		recommended ROE for CMP. As noted above, investors consider the ROE determination
10		by a regulator when assessing the risk of that company as compared to utilities of
11		comparable risk operating in other jurisdictions. The risk premium analysis takes into
12		account this comparison by estimating the return expectations of investors based on the
13		current and past ROE awards of electric utilities across the U.S.
14	VIII.	REGULATORY AND BUSINESS RISKS
15	Q.	Do the DCF, CAPM, and ECAPM results for the proxy group, taken alone, provide
16		an appropriate estimate of the cost of equity for CMP?
17	Α.	No. These results provide only a range of the appropriate estimate of the Company's cost
18		of equity. There are several additional factors that must be taken into consideration when
19		determining where the Company's cost of equity falls within the range of results. These
20		factors, which are discussed below, should be considered with respect to their overall
21		effect on the Company's risk profile.

### A. Capital Expenditures

2	Q.	Please summarize the projected capital expenditure requirements for CMP.
3	А.	The distribution capital expenditure projections for CMP are approximately \$984 million
4		for the period from 2022 through 2026. As discussed in the Capital Investment panel
5		testimony, the Company's capital plan includes expenditures to enhance reliability,
6		increase system resiliency, expand the system to accept load growth resulting from
7		beneficial electrification, increase distributed generation and address aging
8		infrastructure. <sup>68</sup> Based on the Company's net utility plant of approximately \$1,418.82
9		million as of December 31, 2021, the \$984 million anticipated capital expenditures is
10		approximately 69.37% of CMP's net utility plant as of December 31, 2021.
11	Q.	How are the Company's risk profile affected by their substantial capital
12		expenditure requirements?
13	Α.	As with any utility faced with substantial capital expenditure requirements, the
14		Company's risk profile may be adversely affected in two significant and related ways:
15		(1) the heightened level of investment increases the risk of under-recovery or delayed
16		recovery of the invested capital; and (2) an inadequate return would put downward
17		pressure on key credit metrics.
18	Q.	Do credit rating agencies recognize the risks associated with significant capital
19		expenditures?
20	Α.	Yes, they do. From a credit perspective, the additional pressure on cash flows associated
21		with high levels of capital expenditures exerts corresponding pressure on credit metrics

<sup>&</sup>lt;sup>68</sup> Testimony for Capital Investment Panel.

and, therefore, credit ratings. To that point, S&P explains the importance of regulatory
 support for large capital projects:

3 4 5 6 7 8 9 10 11 12 13 14		<ul> <li>Broad support for all capital spending is the most credit-sustaining. Support for only specific types of capital spending, such as specific environmental projects or system integrity plans, is less so, but still favorable for creditors. Allowance of a cash return on construction work-in-progress or similar ratemaking methods historically were extraordinary measures for use in unusual circumstances, but when construction costs are rising, cash flow support could be crucial to maintain credit quality through the spending program. Even more favorable are those jurisdictions that present an opportunity for a higher return on capital projects as an incentive to investors.<sup>69</sup></li> <li>Therefore, to the extent that the Company's rates do not permit the opportunity to recover its capital investments on a regular and timely basis, the Company will face increased recovery risk and thus increased pressure on its credit metrics.</li> </ul>
15	0.	Have you conducted any analysis of the Company's projected capital expenditures
16		relative to the proxy companies?
17	A.	As shown at Exhibit AEB-9, I calculated the ratio of expected capital expenditures to net
18		utility plant for CMP and each of the companies in the proxy group by dividing each
19		company's projected capital expenditures for the period 2022-2026 by its total net utility
20		plant as of December 31, 2021. As shown at Exhibit AEB-9 (see also Figure 14 below),
21		CMP's ratio of capital expenditures as a percentage of net utility plant of 69.37% is
22		approximately 1.30 times the median for the proxy group companies of 53.44%. This
23		result indicates greater risk relative to the companies in the proxy group.

<sup>&</sup>lt;sup>69</sup> S&P Global Ratings, Ratings Direct, "Assessing U.S. Investor-Owned Utility Regulatory Environments," August 10, 2016, at 7.



#### FIGURE 14: COMPARISON OF CAPITAL EXPENDITURES – PROXY GROUP COMPANIES

# Q. What are your conclusions regarding the effect of the Company's capital spending requirements on its risk profile and cost of capital?

3 A. CMP's capital expenditure requirements as a percentage of net utility plant are significant and are proposed to continue over the next few years. Historically, CMP has relied 4 5 entirely on rate case filings for capital cost recovery. In this proceeding, CMP is 6 proposing to recover capital investments through a combination of annual price increases 7 based on forecast plant additions for base projects and programs during each year of the 8 multi-year rate plan and reconciling mechanisms for certain incremental programs. More 9 timely recovery through appropriate rate mechanisms has been relied upon in the industry 10 for significant capital programs. As shown in Exhibit AEB-10, 56.94% of the companies 11 in the proxy group have some form of capital cost recovery mechanisms in place. Therefore, CMP's proposed recovery of capital expenditures is generally consistent with 12 13 the proxy group companies and therefore would not result in any difference in risk when

2

compared to the group. However, to the extent that more timely recovery was not authorized, CMP's relative risk would be greater than the proxy group.

3

#### B. Regulatory Environment

#### 4 Q. Please explain how the regulatory framework affects investors' risk assessments.

5 A. The ratemaking process is premised on the principle that, for investors and companies to 6 commit the capital needed to provide safe and reliable utility services, the subject utility 7 must have the opportunity to recover invested capital and the market-required return on such capital. Regulatory commissions recognize that because utility operations are 8 9 capital intensive, regulatory decisions should enable the utility to attract capital at 10 reasonable terms, which balances the long-term interests of investors and customers. In that respect, the regulatory framework in which a utility operates is one of the most 11 12 important factors considered in both debt and equity investors' risk assessments.

Because investors have many investment alternatives, even within a given market 13 14 sector, the Company's authorized returns must be adequate on a relative basis to ensure 15 their ability to attract capital under a variety of economic and financial market conditions. 16 From the perspective of debt investors, the authorized return should enable the Company 17 to generate the cash flow needed to meet their near-term financial obligations, make the capital investments needed to maintain and expand their systems, and maintain sufficient 18 19 levels of liquidity to fund unexpected events. This financial liquidity must be derived not 20 only from internally generated funds, but also from efficient access to capital markets.

From the perspective of equity investors, the authorized return must be adequate to provide a risk-comparable return on the equity portion of the Company's capital investments. Because equity investors are the residual claimants on the Company's cash flows (that is, debt interest must be paid prior to any equity dividends), equity investors are particularly concerned with the regulatory framework in which a utility operates and
 its effect on future earnings and cash flows.

### Q. Please explain how credit rating agencies consider the regulatory framework in establishing a company's credit rating.

A. Both S&P and Moody's consider the overall regulatory framework in establishing credit
ratings.

7 In the rating agency's evaluation of CMP, Moody's relies on its ratings 8 methodology for Regulated Electric and Gas Networks. The criteria used for networks 9 are based on five key factors: (1) regulatory environment and asset ownership model; (2) 10 scale and complexity of capital program; (3) financial policy; (4) leverage and coverage 11 metrics, and (5) structural considerations and sources of rating uplift from creditor 12 protection. Of these criteria, regulatory environment and asset ownership model is 13 assigned 40.00E. Within that weighting, stability and predictability of regulatory regime 14 and cost and investment recovery (the ability to recover and the timeliness of recovery) 15 are each given a weighting of 15.00%. Leverage and coverage ratios and metrics are also weighted a total of 40.00%. Therefore, Moody's assigns regulatory risk a very 16 significant overall weighting.<sup>70</sup> 17 18 S&P also identifies the regulatory framework as an important factor in credit

ratings for regulated utilities, stating: "One significant aspect of regulatory risk that influences credit quality is the regulatory environment in the jurisdictions in which a

<sup>&</sup>lt;sup>70</sup> Moody's Investors Service, Rating Methodology: Regulated Electric and Gas Networks, March 16, 2017, at 4.

1		utility operates."71 S&P identifies four specific factors that it uses to assess the credit
2		implications of the regulatory environment in which investor-owned regulated utilities
3		operate: (1) regulatory stability; (2) tariff-setting procedures and design; (3) financial
4		stability; and (4) regulatory independence and insulation. <sup>72</sup>
5	Q.	How does the regulatory environment in which a utility operates affect its access to
6		and cost of capital?
7	A.	The regulatory environment can significantly affect both the access to, and cost of,
8		capital in several ways. As noted by Moody's:
9 10		We consider the characteristics of the regulatory environment in which a network operates. These include how developed and transparent the regulatory
11		framework is: the strength of the political and legal underpinnings of the
12		regulatory framework; the regulator's track record for predictability and
13		stability in terms of decision making; its independence from political
14		interference; and our forward looking view on whether these conditions will
15		continue to persist. In addition, this sub-factor also considers the effectiveness
16		of the independent body or legal system that can arbitrate disputes between a
17		regulator and a regulated company in a timely fashion. <sup>73</sup>
18		Moody's further highlighted the relevance of a stable and predictable regulatory
19		environment to a utility's credit quality, noting: "the predictability and supportiveness of
20		the regulatory framework in which a network operates - as well as the legal and political
21		framework that underpins it – is a key credit consideration and one that differentiates this
22		sector from most other corporate sectors."74

<sup>&</sup>lt;sup>71</sup> Standard & Poor's Global Ratings, Ratings Direct, U.S. and Canadian Regulatory Jurisdictions Support Utilities' Credit Quality—But Some More So Than Others, June 25, 2018, at 2.

 $<sup>^{72}</sup> Id.$  at 1.

<sup>&</sup>lt;sup>73</sup> Moody's Investors Service, Rating Methodology: Regulated Electric and Gas Networks, March 16, 2017, at 8.

<sup>&</sup>lt;sup>74</sup> Id. at 7
1	Q.	Have you conducted any analysis of the regulatory framework in Maine relative to
2		the jurisdictions in which the companies in your proxy group operate?

3 Yes. I have evaluated the regulatory framework in Maine considering two factors which Α. 4 are important to ensuring CMP maintains access to capital at reasonable terms. As I will 5 discuss in more detail below, the two factors are: (1) cost recovery mechanisms which 6 allow a utility to recover costs in a timely manner between rate cases and provide the 7 utility the opportunity to earn its authorized return; and (2) comparable return standard<sup>75</sup> because an awarded ROE that is significantly below the ROEs awarded to other utilities 8 9 with comparable risks can affect the ability of a utility to attract capital at reasonable 10 terms.

11

#### Cost Recovery Mechanisms

1.

Q. Have you conducted any analysis to compare the cost recovery mechanisms of CMP
 to the cost recovery mechanisms approved in the jurisdictions in which the companies
 in your proxy group operate?

15 Yes. I selected four mechanisms that are important to provide a regulated utility an Α. 16 opportunity to earn its authorized ROE. These are: (1) test year convention (*i.e.*, forecast 17 vs. historical); (2) method for determining rate base (*i.e.*, average vs. year-end); (3) use of revenue decoupling mechanisms or formula-based rates that mitigate volumetric risk; and 18 19 (4) prevalence of capital cost recovery between rate cases. The results of this cost 20recovery assessment are shown in Exhibit AEB-10 and are summarized below: 21 • Test year convention: CMP is proposing a forward looking rate plan that escalates

22

from the test-year to each rate year and a multi-year capital plan. However, the

<sup>&</sup>lt;sup>75</sup> *Hope* and *Bluefield* require the return be commensurate with returns on investments in enterprises with similar risk.

1		Commission has authorized a historical test-year escalated to the rate year based on
2		the 5-year compound annual growth in plant additions and the escalation of O&M
3		costs to the rate year. As shown in Exhibit AEB-10, 51.39% of the proxy group
4		provide service in jurisdictions that use a fully or partially forecast test year. Forecast
5		test years have been relied on for several years and produce cost estimates that are
6		more reflective of future costs which results in more accurate recovery of incurred
7		costs and mitigates the regulatory lag associated with historical test years.
8	•	Rate base: CMP's rate base in Maine is determined based on the average of the
9		beginning and ending test year rate base balances, while as shown in Exhibit AEB-10,
10		44.44% of the operating companies held by proxy group relied on a year-end rate
11		base as the test year. The year-end methodology means that the rate base includes
12		capital additions that occurred in the second half of the test year and is more
13		reflective of total net utility plant going forward.
14	•	Non-Volumetric Rate Design: CMP does have protection against volumetric risk in
15		Maine, through a revenue decoupling mechanism that was approved for continuation
16		in the Company's last rate case. Similarly, 89 out of 144 (61.81%) of the operating
17		companies held by the proxy group have non-volumetric rate design through either
18		straight fixed variable rate design, revenue decoupling mechanisms or formula rate
19		plans that allow them to break the link between customer usage and revenues.
20	•	Capital Cost Recovery: As discussed above, CMP is proposing to recover capital
21		investments through a combination of annual price increases based on forecast plant
22		additions for base projects and programs during each year of the multi-year rate plan
23		and reconciling mechanisms for certain incremental programs. As shown in Exhibit

# **ROE-62**

1		AEB-10, 56.94% of the operating companies held by the proxy group have some
2		form of capital cost recovery mechanism in place.
3		2. Authorized ROEs
4	Q.	How do recent returns in Maine compare to the authorized returns in other
5		jurisdictions?
6	Α.	The Commission has historically relied primarily on the results of the DCF analysis with
7		results of the CAPM model used as a check on the reasonableness of the DCF results.
8		This practice has reduced the authorized ROE for electric utility operations in Maine.
9		Figure 15 below shows the authorized returns for electric utilities in other jurisdictions
10		since January 2009, and the returns authorized in Maine for electric utilities. As shown
11		in Figure 15, the authorized returns for electric utilities in Maine have been below the
12		average authorized ROE for electric utilities in other jurisdictions since 2009.



## FIGURE 15: COMPARISON OF MAINE AND U.S. AUTHORIZED ELECTRIC RETURNS<sup>76</sup>

#### 1 Q. Should the Commission be concerned about authorizing equity returns that are at 2 the low end of the range established by other state regulatory jurisdictions? 3 Α, Yes. Placing CMP at the low end of authorized ROEs outside Maine over the longer term can negatively affect the Company's access to capital and the overall cost of capital. 4 5 As I discuss below, the recent negative rate case determination, including a below 6 average authorized ROE, for Arizona Public Service resulted in a 24% decline in the 7 share price for Pinnacle West Capital, increasing the overall cost of equity for that 8 company.

<sup>&</sup>lt;sup>76</sup> S&P Capital IQ Pro. Electric rate case decisions from January 1, 2009 through May 31, 2022. The chart does not display the 12.88% ROE that was authorized for Alaska Electric Light and Power on September 2, 2011. The chart also excludes the authorized returns in Illinois and Vermont since they are established based on a formulaic approach that is directly linked to interest rates and therefore is affected by market conditions and monetary policy. Finally, the chart excludes the authorized ROE for CMP in the Company's last rate case of 8.25% because the authorized return included a 100 basis point ROE penalty.

1		Second, as noted in Sections V and VII, interest rates are expected to increase as
2		the Federal Reserve normalizes monetary policy, and thus utilities are expected to
3		underperform over the near-term. If utility stocks underperform over the near-term then
4		utility dividend yields will increase resulting in higher estimates of the ROE results
5		produced by the DCF model. Therefore, the results of the DCF model using current
6		market information will underestimate investors' expected ROE over the time period in
7		which CMP's rates will be in effect. As a result, it is important that the Commission
8		consider, the results of alternative methods such as the forward looking CAPM, ECAPM,
9		and Bond Yield Plus Risk Premium and the returns that have been authorized by other
10		electric utilities across the U.S.
11	Q.	Do credit rating agencies consider the authorized ROE in the overall risk
12		assessment of a utility?
13	Α.	Yes, they do. To the extent that the returns in a jurisdiction are lower than the returns
13 14	Α.	Yes, they do. To the extent that the returns in a jurisdiction are lower than the returns that have been authorized more broadly, credit rating agencies will consider this in the
13 14 15	A.	Yes, they do. To the extent that the returns in a jurisdiction are lower than the returns that have been authorized more broadly, credit rating agencies will consider this in the overall risk assessment of the regulatory jurisdiction in which the company operates. It is
13 14 15 16	А.	Yes, they do. To the extent that the returns in a jurisdiction are lower than the returns that have been authorized more broadly, credit rating agencies will consider this in the overall risk assessment of the regulatory jurisdiction in which the company operates. It is important to consider credit ratings because they affect the overall cost of borrowing, and
13 14 15 16 17	A.	Yes, they do. To the extent that the returns in a jurisdiction are lower than the returns that have been authorized more broadly, credit rating agencies will consider this in the overall risk assessment of the regulatory jurisdiction in which the company operates. It is important to consider credit ratings because they affect the overall cost of borrowing, and they act as a signal to equity investors about the risk of investing in the equity of a
13 14 15 16 17 18	Α.	<ul> <li>Yes, they do. To the extent that the returns in a jurisdiction are lower than the returns</li> <li>that have been authorized more broadly, credit rating agencies will consider this in the</li> <li>overall risk assessment of the regulatory jurisdiction in which the company operates. It is</li> <li>important to consider credit ratings because they affect the overall cost of borrowing, and</li> <li>they act as a signal to equity investors about the risk of investing in the equity of a</li> <li>company. Therefore, lower credit ratings can affect both the cost of debt and equity.</li> </ul>
<ol> <li>13</li> <li>14</li> <li>15</li> <li>16</li> <li>17</li> <li>18</li> <li>19</li> </ol>	Α.	<ul> <li>Yes, they do. To the extent that the returns in a jurisdiction are lower than the returns</li> <li>that have been authorized more broadly, credit rating agencies will consider this in the</li> <li>overall risk assessment of the regulatory jurisdiction in which the company operates. It is</li> <li>important to consider credit ratings because they affect the overall cost of borrowing, and</li> <li>they act as a signal to equity investors about the risk of investing in the equity of a</li> <li>company. Therefore, lower credit ratings can affect both the cost of debt and equity.</li> <li>Examples of recent credit rating agency responses include ALLETE, Inc., CenterPoint</li> </ul>
<ol> <li>13</li> <li>14</li> <li>15</li> <li>16</li> <li>17</li> <li>18</li> <li>19</li> <li>20</li> </ol>	A.	<ul> <li>Yes, they do. To the extent that the returns in a jurisdiction are lower than the returns</li> <li>that have been authorized more broadly, credit rating agencies will consider this in the</li> <li>overall risk assessment of the regulatory jurisdiction in which the company operates. It is</li> <li>important to consider credit ratings because they affect the overall cost of borrowing, and</li> <li>they act as a signal to equity investors about the risk of investing in the equity of a</li> <li>company. Therefore, lower credit ratings can affect both the cost of debt and equity.</li> <li>Examples of recent credit rating agency responses include ALLETE, Inc., CenterPoint</li> <li>Energy Houston Electric and Pinnacle West Capital Corporation. Moody's downgraded</li> </ul>
<ol> <li>13</li> <li>14</li> <li>15</li> <li>16</li> <li>17</li> <li>18</li> <li>19</li> <li>20</li> <li>21</li> </ol>	A.	Yes, they do. To the extent that the returns in a jurisdiction are lower than the returns that have been authorized more broadly, credit rating agencies will consider this in the overall risk assessment of the regulatory jurisdiction in which the company operates. It is important to consider credit ratings because they affect the overall cost of borrowing, and they act as a signal to equity investors about the risk of investing in the equity of a company. Therefore, lower credit ratings can affect both the cost of debt and equity. Examples of recent credit rating agency responses include ALLETE, Inc., CenterPoint Energy Houston Electric and Pinnacle West Capital Corporation. Moody's downgraded ALLETE, Inc. from A3 to Baa1 primarily based on the less than favorable outcome in

1	Moody's noted was a below average authorized ROE of 9.25%.77 In addition,
2	FitchRatings downgraded CenterPoint Energy Houston Electric's ("CEHE") Long-Term
3	Issuer Default rating from A- to BBB+ and revised the rating outlook from Stable to
4	Negative following the approval of an unfavorable outcome in a recent rate case in
5	Texas. <sup>78</sup> Finally, FitchRatings recently downgraded and maintained a negative outlook
6	for Arizona Public Service Company ("APS") and its parent, Pinnacle West Capital
7	Corporation, following the hearings conducted by the Arizona Corporation Commission
8	("ACC") in October 2021 regarding APS' current rate case proceeding. <sup>79</sup> While the ACC
9	had not issued a final order in APS' rate case at the time, FitchRatings noted that the
10	developments at the hearing in October indicate a likely credit negative outcome that will
11	negatively affect the financial metrics of both APS and Pinnacle West Capital
12	Corporation. It is also important to note that both Standard & Poor's and Moody's
13	downgraded Pinnacle West Capital's and APS' credit rating and put the companies on
14	credit watch negative following the Commission's November vote that officially
15	authorized the 8.70% ROE. <sup>80</sup>

<sup>&</sup>lt;sup>77</sup> Moody's Investors Service, "Credit Opinion: ALLETE, Inc. Update following downgrade," at 3 (April 3, 2019).

<sup>&</sup>lt;sup>78</sup> FitchRatings, "Fitch Downgrades CenterPoint Energy Houston Electric to BBB+; Affirms CNP; Outlooks Negative," February 19, 2020.

<sup>&</sup>lt;sup>79</sup> FitchRatings, "Fitch Downgrades Pinnacle West Capital & Arizona Public Service to 'BBB+'; Outlooks Remain Negative," October 12, 2021.

<sup>&</sup>lt;sup>80</sup> See S&P Capital IQ and Moody's Investors Service, "Rating Actions: Moody's downgrades Pinnacle West to Baa1 and Arizona Public Service to A3; outlook negative," (Nov. 17, 2021).

1	Q.	Are you aware of any utilities whose market data has been affected by adverse rate
2		case developments?
3	A.	Yes, I am. The market has responded negatively to recent returns authorized by the
4		ACC. As noted above, the most recent ROE determination in Arizona was for APS. The
5		Recommended Opinion and Order ("ROO") issued in the APS rate proceeding on
6		August 2, 2021, recommended an ROE of 9.16%. In October 2021, that recommendation
7		was amended to reduce the company's ROE to 8.70%. <sup>81</sup> The final ROE that was
8		established for APS was 8.70%. The market reacted strongly to the proposed order and
9		subsequent amendment and final decision. Guggenheim Securities LLC, an equity
10		analyst that follows Pinnacle West Capital Corporation, the parent company of APS,
11		informed its clients that:
12 13 14		[T]he "Arizona Corporation Commission is now confirmed to be the single most value destructive regulatory environment in the country as far as investor-owned utilities are concerned". <sup>82</sup>
15		S&P Global Market Intelligence (Regulatory Research Associates) noted that this
16		decision was "among the lowest ROEs RRA had encountered in its coverage of vertically
17		integrated electric utilities in the past 30 years."83
18		As shown in Figure 16 below, Pinnacle West Capital Corporation's stock price
19		declined approximately 24% from August 2, 2021 to November 4, 2021 following the
20		issuance of the ROO, which recommended an ROE of 9.16%, and then the subsequent
21		amendment to that opinion recommending the 8.70% ROE ultimately adopted by the

<sup>&</sup>lt;sup>81</sup> Arizona Corporation Commission Docket No. E-01345A-19-0236, Commissioner Olson Proposed Amendment No. 1 to the Recommended Opinion and Order. October 4, 2021.

<sup>&</sup>lt;sup>82</sup> S&P Global Market Intelligence, "Pinnacle West shares tumble after regulators slash returns in rate case," October 7, 2021.

<sup>&</sup>lt;sup>83</sup> S&P Global Market Intelligence, RRA Regulatory Focus, "Commission accords Arizona Public Service Company a well below average ROE," October 8, 2021.

ACC. Moreover, the Value Line five-year projected EPS growth rates for this company have fallen from 5.0% in July 2021, prior to the deliberations in the rate proceeding to "Nil" in October 2021 and most recently 1.5% in April 2022. This recent Value Line report noted that PNW's earnings would "almost certainly decline in 2022" primarily related to the APS rate order. For PNW, the APS decision has had a significant effect on the share price and growth rate assumptions used in the DCF model.

FIGURE 16: PINNACLE WEST CAPITAL STOCK PRICE VS. S&P 500 UTILITIES



Q. How should the Commission use the information regarding authorized ROEs in
other jurisdictions in determining the ROE for CMP?

- 9 A. As discussed above, the companies in the proxy group operate in multiple jurisdictions
- 10 across the U.S. Since CMP must compete directly for capital with investments of similar
- 11 risk, it is appropriate to review the authorized ROEs in other jurisdictions. The

1		comparison is important because investors are considering the authorized returns across
2		the U.S. and are likely to invest equity in those utilities with the highest returns.
3		3. Regulatory Risk
4	Q.	Have you developed any additional analyses to evaluate the regulatory environment
5		in Maine as compared to the jurisdictions in which the companies in your proxy
6		group operate?
7	A.	Yes. I have conducted two additional analyses to compare the regulatory framework of
8		Maine to the jurisdictions in which the companies in the proxy group operate.
9		Specifically, I considered two different rankings: (1) the Regulatory Research Associates
10		("RRA") ranking of regulatory jurisdictions; and (2) S&P's ranking of the credit
11		supportiveness of regulatory jurisdictions.
12	Q.	Please explain how you used the RRA ratings to compare the regulatory
13		jurisdictions of the proxy companies with the Company's regulatory jurisdiction.
14	Α.	RRA develops its ranking based on its assessment of how investors perceive the
15		regulatory risk associated with ownership of utility securities in that jurisdiction,
16		specifically reflecting its assessment of the probable level and quality of earnings to be
17		realized by the State's utilities as a result of regulatory, legislative, and court actions.
18		RRA assigns a ranking for each regulatory jurisdiction between "Above Average/1" to
19		"Below Average/3," with nine total rankings between these categories. I applied a
20		numeric ranking system to the RRA rankings with "Above Average/1" assigned the
21		highest ranking ("1") and "Below Average/3" assigned the lowest ranking ("9"). As
22		shown on Exhibit AEB-11, the Maine jurisdictional ranking ("Average/3" - "6.0") was
23		below the proxy group average ranking ("Average/1 – Average/2" – "4.63") from RRA.

# 1 Q. Has RRA provided recent commentary regarding its regulatory ranking for the

- 2 Maine?
- 3 A. Yes, they have. In fact, in March 2020, RRA downgraded the regulatory environment
- 4

ranking for Maine from Average/2 to Average/3. RRA recently noted the following:

Maine regulation of energy utilities is somewhat restrictive from an investor 5 6 viewpoint. The PUC's ROE determinations, historically, have been somewhat 7 below prevailing industry averages when established. In a rate case for Central 8 Maine Power, or CMP, the PUC imposed a penalty to reflect "imprudent" 9 management decisions with respect to a new billing system. The penalty 10 reduced the utility's authorized ROE below the average of ROEs authorized by state commission in cases decided in 2019, and was the lowest equity return 11 12 authorization for an electric utility nationwide since RRA began tracking 13 equity returns in the 1980s. The penalty was removed in February 2022. The PUC has generally relied on an average original-cost rate base for an historical 14 test period adjusted for known-and-measurable changes, which tends to 15 contribute to regulatory lag and makes it difficult for the utilities to earn the 16 authorized return. Full decoupling mechanisms are in place for the electric 17 18 utilities, a mechanism is also in place for one electric utility that reflects costs 19 of significant weather events and one of the state's gas utilities uses a targeted 20 infrastructure replacement adjustment mechanism that provides for recovery of 21 certain operational and safety-related infrastructure replacement and upgrade 22 projects. The state took a constructive approach to electric industry restructuring. Full electric retail competition has been in place since 2000. 23 The investor-owned utilities were required to divest most of their non-nuclear 24 25 generation assets, and are being accorded a reasonable opportunity to recover stranded costs, albeit over an extended timeframe. The power to meet 26 standard-offer-service requirements is procured through a PUC-administered 27 28 competitive bid and is provided by third parties; therefore, the utilities have no exposure to market-price risk. Historically, natural gas service has been 29 30 limited in the state, and over the last several years, the commission has adopted 31 innovative pricing and regulatory mechanisms to encourage gas industry development. Cost-of-gas adjustment mechanisms are utilized by the state's 32 gas utilities. Legislation enacted in 2019 amended the PUC's standard of 33 34 approval for public utility corporate reorganizations to a "net benefits" standard from a "no net harm" standard. RRA accords Maine regulation an 35 Average/3 ranking.84 36

<sup>&</sup>lt;sup>84</sup> Regulatory Research Associates, Profile of Maine Public Utilities Commission, accessed June 30, 2022.

1

**Q**.

#### How did you conduct your analysis of the S&P credit supportiveness?

2 Α. For credit supportiveness, S&P classifies each regulatory jurisdiction into five categories 3 that range from "Credit Supportive" to "Most Credit Supportive." My analysis of the 4 credit supportiveness of the regulatory jurisdictions that the proxy companies operate in, 5 as compared with the Company's regulatory jurisdiction, was similar to the analysis of 6 the RRA overall regulatory ranking discussed above. I assigned a numerical ranking to 7 each category, from Most Credit Supportive ("1") to Credit Supportive ("5"). As shown 8 in Exhibit AEB-12, the proxy group average ranking was 2.43, which would be classified 9 between "Highly Credit Supportive" and "Very Credit Supportive", and is generally 10 consistent with the Maine jurisdictional classification of "Highly Credit Supportive" 11 ("2"). 12 What is your conclusion regarding the regulatory framework in Maine as compared **Q**. 13 with the jurisdictions in which the proxy group companies operate? 14 Α. As discussed throughout this section of my testimony, both Moody's and S&P have 15 identified the supportiveness of the regulatory environment as an important consideration in developing their overall credit ratings for regulated utilities. Considering the 16 17 regulatory adjustment mechanisms, many of the companies in the proxy group have 18 timely cost recovery through forecasted test years, year-end rate base, cost recovery 19 trackers and revenue stabilization mechanisms. Additionally, authorized ROEs in Maine 20have been below the average authorized ROEs for electric utilities across the U.S. 21 Moreover, RRA recently downgraded the RRA jurisdictional ranking for Maine; thus a 22 comparison of Maine's RRA jurisdictional ranking to the proxy group indicates greater risk than the average for the proxy group. For these reasons, I conclude that CMP has 23

1		greater than average regulatory risk when compared to the proxy group, indicating that
2		the authorized ROE for CMP should be higher than the proxy group median.
3	IX.	CAPITAL STRUCTURE
4	Q.	Is the capital structure of the Company an important consideration in the
5		determination of the appropriate ROE?
6	Α.	Yes, it is. Assuming other factors are equal, a higher debt ratio increases the risk to
7		investors. For debt holders, higher debt ratios result in a greater portion of the available
8		cash flow being required to meet debt service, thereby increasing the risk associated with
9		the payments on debt. The result of increased risk is a higher interest rate. The
10		incremental risk of a higher debt ratio is more significant for common equity
11		shareholders, who are the residual claimants on the cash flow of the Company.
12		Therefore, the greater the debt service requirement, the less cash flow is available for
13		common equity holders.
14	Q.	What is CMP's proposed capital structure?
15	Α.	The Company is proposing a hypothetical capital structure consisting of 50.00% common
16		equity, 48.87% long-term debt, 0.02% preferred equity, and 1.11% short-term debt.
17	Q.	Is the Company's proposed hypothetical equity ratio consistent with the Company's
18		actual equity ratio?
19	А.	No, it is not. As shown in Figure 17 below, CMP's actual equity ratio has ranged from
20		53.60% to 59.57% over the period from 2013-2021. As a result, the Company's
21		proposed hypothetical capital structure is well below the actual equity ratio of the
22		Company.

Year	Year End Equity Ratio
2013	54.66%
2014	53,60%
2015	55,27%
2016	57.67%
2017	59.57%
2018	59.08%
2019	58,58%
2020	57.88%
2021	58.97%
Average	57.25%

# FIGURE 17: HISTORICAL CMP EQUITY RATIO

1 Q. Why is the Company proposing that a hypothetical capital structure be used for 2 ratemaking purposes as opposed to the Company's actual capital structure? 3 A. The Commission has historically relied on the use of a hypothetical capital structure consisting of 50% common equity for CMP.<sup>85</sup> Therefore, the Company has proposed a 4 5 hypothetical capital structure that is consistent with the Commission's prior decisions for 6 the Company. 7 Q. Did you conduct any analysis to determine if this hypothetical equity ratio was 8 reasonable? 9 Yes, I did. I reviewed the Company's proposed capital structure and the capital Α.

- 10 structures of the utility operating subsidiaries of the proxy companies. Because the ROE
- 11 is set based on the return that is derived from the risk-comparable proxy group, it is
- 12 reasonable to look to the proxy group average capital structure to benchmark the equity
- 13 ratio for the Company.

<sup>&</sup>lt;sup>85</sup> CMP, Investigation into Rates and Revenue Requirements of Central Maine Power Company, Docket No. 2018-00194, Order, (February 19, 2020) at 87.

A. I calculated the mean proportions of common equity and long-term debt for the most
recent four years (2018-2021) for each of the companies in the proxy group at the
operating subsidiary level. My analysis of the capital structures of the proxy group
companies is provided in Exhibit AEB-13. As shown in Exhibit AEB-13, the equity
ratios for the proxy group ranged from 44.95% to 63.58%, with an average of 54.10%.
CMP's proposed equity ratio of 50.00% is well below the average equity ratio for the
utility operating subsidiaries of the proxy group.

Please discuss your analysis of the capital structures of the proxy group companies.

9

1

**Q**.

#### Q. Why is it appropriate to consider the equity ratio for the proxy companies?

10 Α. As noted above, the determination of the ROE is based on the expected return for a proxy 11 group of companies that are comparable in risk to CMP. The equity ratio is a measure of 12 the financial risk of the company, and the authorized ROE is the return to compensate 13 investors for that risk. If the Commission is going to rely on the ROE estimates for the 14 proxy companies to establish the authorized ROE for CMP, it is important that the 15 financial risk of CMP be similar to the financial risk of the proxy group. This is accomplished when the equity ratio of the subject company (in this case CMP) is 16 17 comparable to the equity ratios of the proxy group.

18 Q. Have you reviewed any academic literature that provides support for the use of
 19 capital structures of the proxy group companies in determining the reasonableness
 20 of the subject company's capital structure?

A. Yes, I have. Dr. Morin noted that the use of a hypothetical capital structure is only
appropriate if the estimated cost of equity based on current investor expectations is
adjusted to account for the change from the company's actual capital structure (*i.e.*, the

1	capital structure expected by investors) to the hypothetical capital structure. <sup>86</sup> As noted
2	above, in the current proceeding the cost of equity for CMP is being estimated based on
3	the expected return for a proxy group of companies that are of comparable risk to CMP.
4	Therefore, the estimated cost of equity is based on the financial risk of the proxy group
5	which is in turn measured by the capital structure. If the company's equity ratio were to
6	deviate from the proxy group then an adjustment to the cost of equity would be needed to
7	account for the difference in financial risk. Dr. Morin noted the importance of adjusting
8	for the difference in financial risk:
9	[i]f a regulator were to impute a capital structure consisting of substantially

10 more (less) debt than the rate year capital structure, the higher (lower) common equity cost rate related to the changed common equity ratio should be reflected 11 in the approach. If the regulator ascribes a capital structure different from the 12 test year capital structure, which imputes a higher debt amount for example, 13 14 the repercussions on equity costs must be recognized. It is a rudimentary tenet 15 of basic finance that the greater the amount of financial risk borne by common 16 shareholders, the greater the return required by shareholders in order to be 17 compensated for the added financial risk imparted by the greater use of debt financing. In other words, the greater the debt ratio, the greater is the return 18 19 required by equity investors. Both the cost of incremental debt and the cost of 20 equity must be adjusted to reflect the additional risk associated with the more 21 debt-heavy capital structure. Lower common equity ratios imply greater risk and higher capital cost, and conversely.87 22

# 23 Q. Would the use of the Company's proposed 50% hypothetical equity ratio have any

- 24 implications for the Company's ROE?
- 25 A. Yes. The average equity ratio of the proxy companies is well above 50.00%, which
- 26 means that, all else equal, the proxy companies have lower financial risk than is implied
- by the Company's hypothetical 50.00% equity ratio. Given this risk differential and the
- 28 significance of the overall ROE/capital structure determination to the Company's

<sup>&</sup>lt;sup>86</sup> Dr. Roger Morin, Modern Regulatory Finance, 2021, at 541

<sup>&</sup>lt;sup>87</sup> Dr. Roger Morin, Modern Regulatory Finance, 2021, at 541

continuing ability to access capital, the use of a hypothetical capital structure consisting
 of 50% equity would support an ROE towards the higher-end of my recommended ROE
 range.

4 О. Are there other factors to be considered in setting the Company's capital structure? 5 A. Yes. The credit rating agencies' response to the Tax Cuts and Jobs Act of 2017 (TCJA) 6 must also be considered when determining the equity ratio. All three rating agencies 7 have noted that the TCJA has negative implications for utility cash flows. S&P and Fitch specifically identified increasing the equity ratio as one approach to ensure that utilities 8 9 have sufficient cash flows following the federal income tax rate reductions and the loss of 10 bonus depreciation. As S&P noted "[r]egulators must also recognize that tax reform is a strain on utility credit quality, and we expect companies to request stronger capital 11 structures and other means to offset some of the negative impact."88 Furthermore. 12 13 Moody's downgraded the rating outlook for the entire utilities sector in June 2018 and 14 has continued to downgrade the ratings of utilities based in part on the negative effects of 15 the TCJA on cash flows. 16 S&P continues to maintain a negative outlook for the utility industry in 2022 and 17 noted that since downgrades outpaced upgrades for a second consecutive year in 2021 for the first time ever the median investor-owned utility credit rating fell to the "BBB" 18

19 category.<sup>89</sup> Further, S&P expects continued pressure on cash flows over the near-term as

- 20 utilities continue to increase leverage to fund capital expenditure plans necessary to
- 21

reduce greenhouse gas emission and improve safety and reliability. Finally, S&P also

\_\_\_\_\_

<sup>&</sup>lt;sup>88</sup> Standard & Poor's Ratings, "U.S. Tax Reform: For Utilities' Credit Quality, Challenges Abound," January 24, 2018, at 5.

<sup>&</sup>lt;sup>89</sup> S&P Global Ratings, "For the First Time Ever, the Median Investor-Owned Utility Ratings Falls to the 'BBB' Category," January 20, 2022.

1		highlighted inflation, higher interest rates and rising commodity prices as additional risks
2		that could further constrain the credit metrics for utilities over the near-term. In regards
3		to inflation S&P noted:
4 5 7 8 9 10 11 12		Inflation recently spiked to its highest level in decades after rising for several consecutive months in 2021. Given the sustained increase to the U.S. consumer price index in 2021, inflation no longer appears to be just transitory and may have financial implications for the investor-owned North American regulated utility industry. Because of the regulatory lag within the industry, inflation, which causes prices to rise, typically leads to a weakening of financial performance. The regulatory lag is the timing difference between when costs are incurred and when regulators allow those costs to be fully recovered from ratepayers. <sup>90</sup>
13		The credit ratings agencies' continued concerns over the negative effects of the TCJA,
14		inflation, and increased capital expenditures underscore the importance of maintaining
15		adequate cash flow metrics for the industry, as a whole, and CMP, particularly, in the
16		context of this proceeding.
17	Q.	What is your conclusion regarding CMP's proposed equity ratio?
18	Α.	Considering the actual capital structures of the proxy group operating companies, I
19		believe that CMP's proposed hypothetical common equity ratio of 50.00% is reasonable.
20		The proposed equity ratio is well below the average equity ratio established by the capital
21		structures of the utility operating subsidiaries of the proxy companies, which would
22		suggest that CMP has greater financial risk than the proxy group. This proposed capital
23		structure would support an ROE towards the high-end of my recommended ROE range.
24		Finally, based on the cash flow concerns raised by credit rating agencies as a result of the
25		TCJA, inflation, and increased capital expenditures, it is reasonable to rely on a higher
26		equity ratio than the Company may have relied on in prior rate cases.

1

2

Q.

#### X. MULTI-YEAR RATE PLAN

#### 3 As discussed in the Policy Panel testimony, the Company is proposing a multi-year rate Α. 4 plan to support the capital investment plan required to build the smarter, stronger and 5 more resilient grid needed to provide safe and reliable service going forward in the face 6 of increasing extreme weather events, aging plant and additional customer expectations 7 and to meet the State of Maine's clean energy and climate policies. 8 Q. Have current market conditions increased the risk of a multi-year stay-out for the 9 **Company?** 10 Α. Yes. While a multi-year plan provides a solution to achieve these objectives more efficiently than filing more frequent rate cases, current market conditions have increased 11 12 the risk of a multi-year stay-out for the Company. As noted earlier in my testimony, 13 inflation is at the one of the highest level seen since 1981; 9.0% in June 2022. In 14 response, the Federal Reserve is currently normalizing monetary policy which will result 15 in increases in long-term interest rates over the near-term. As a result, in the current 16 market environment, there is additional risk that the authorized ROE for the latter years 17 of a multi-year rate plan will be lower than investors' future requirements as interest rates 18 are expected to increase. 19 Q. Have you conducted any analysis as to the expected return on equity over the period 20that the Company is proposing for its multi-year rate plan? 21 Yes. As discussed in Section VII of my testimony, the CAPM and ECAPM analysis Α. 22 shown in Exhibit AEB-5 estimates the projected yields on 30-year Treasury bonds over 23 the period from 2024 - 2028 as the risk-free rate and results in an estimated cost of equity 24 range of 10.41% to 11.92%.

Are you aware that the Company is filing a multi-year rate plan?

#### **ROE-78**

1	Moreover, as also discussed in Section VII, it is possible to use the Bond Yield
2	Plus Risk Premium analysis to estimate the historical relationship between bond yields
3	and the market risk premium. As shown in Exhibit AEB-8, the regression equation
4	indicates that for every 100 basis point change in the yield on 30-year treasury bonds, the
5	risk premium moves in an opposite direction by about 55 basis points. This relationship
6	implies that for every 100 basis point change in Treasury rates, the ROE will change in
7	the same direction by about 45 (100-55) basis points. Considering the projected 30-year
8	Treasury bond yield over the period from 2024–2028, the Bond Yield Plus Risk Premium
9	produces an ROE estimate of 10.18% which is 35 basis points greater than the estimate of
10	the ROE of 9.83% produced by the Bond Yield Plus Risk Premium Analysis relying on
11	the current average of the 30-year Treasury bond yield.

# 12 13

Q.

# What are your conclusions regarding the Company's proposed multi-year rate plan and its effect on the ROE for CMP?

14 Α. Inflation is currently at its highest level seen in approximately 40 years and is expected to 15 remain elevated over the near-term. Thus, investors expect interest rates to increase over 16 the near-term. Because the share prices of utilities are inversely correlated with interest 17 rates, the utility sector is expected to underperform and therefore, estimates of the ROE using current market data are likely to understate the ROE during CMP's proposed multi-18 19 year rate period. As a result, the Commission should adopt an ROE which is based on 20 the use of forward looking data to ensure the Company can attract capital on reasonable 21 terms, under varying market conditions, during the period that rates will be in effect. The Company is requesting an ROE of 10.20% which is below the midpoint of my 22 23 recommended ROE range and therefore a conservative estimate of the investor-required 24 return over the three-year multi-year rate period. However, as the Company explains

**ROE-79** 

1483

elsewhere in testimony, the selection of 10.20% is intended to be sensitive to current
 economic circumstances affecting its customers at this time.

3 XI. CONCLUSIONS AND RECOMMENDATIONS

# 4 Q. What is your conclusion regarding a fair ROE for CMP?

5 Figure 18 below provides a summary of my analytical results for the proxy group. Based A. 6 on these results, it is my view that a reasonable ROE for the Company is within the range 7 of 9.75% to 11.25%. Considering the qualitative analyses presented in my testimony, the 8 business and financial risks of CMP compared to the proxy group, current conditions in 9 capital markets including the expectation for rising interest rates and increasing 10 inflationary pressure, and the Company's proposed capital structure, I conclude that the 11 ROE would reasonably be set above the midpoint of this range. However, the Company 12 is requesting an ROE of 10.20%, which is below the midpoint of my recommended 13 range, in order to take into consideration the effects of currently high inflationary 14 pressures on customers.

Constant Growth DCF				
	Mean Low	Mcan	Mean High	
30-Day Average	8,06%	9,17%	10.28%	
90-Day Average	8,12%	9.24%	10.35%	
180-Day Average	8,24%	9,35%	10.47%	
	Median Low	Median	Median High	
30-Day Average	7.94%	9.07%	9.80%	
90-Day Average	8.03%	9.11%	9.91%	
180-Day Average	8,30%	9.21%	10.06%	
	CAPM			
	Current 30-day Average Treasury Bond Yield	Near-Term Blue Chip Forecast Yicld	Long-Term Blue Chip Forecast Yield	
Value Line Beta	11.50%	11.57%	11,61%	
Bloomberg Beta	10.79%	10.89%	10,95%	
Long-Term Avg. Beta	10.20%	10.33%	10.41%	
	ЕСАРМ			
	Current 30-day Average Treasury Bond Yield	Near-Term Blue Chip Forecast Yield	Long-Term Blue Chip Forecast Yield	
Value Line Beta	11.84%	11.89%	11.92%	
Bloomberg Beta	11.31%	11.38%	11.43%	
Long-Term Avg. Beta	10.87%	10,96%	11,03%	
Bond Yield Plus Risk Premium				
	Current 30-day Average Treasury Bond Yield	Near-Term Blue Chip Forecast Yicld	Long-Term Blue Chip Forecast Yield	
<b>Risk Premium Results</b>	9.83%	10.04%	10.18%	

FIGURE 18: SUMMARY OF ANALYTICAL RESULTS

#### 1 Q. What is your conclusion regarding the Company's proposed common equity ratio?

2 A. I conclude that CMP's proposed hypothetical rate-making capital structure composed of

3 50.00% common equity, 48.87% long-term debt, 0.02% preferred equity, and 1.11%

- 4 short-term debt is reasonable and should be adopted. This conclusion is supported by a
- 5 review of the proxy group capital structures, inflation, the ongoing effect of the TCJA,
- 6 and increased capital expenditures on cash flows.

# 7 Q. Does this conclude your testimony at this time?

8 A. Yes, it does.

Massachusetts Electric Company Nantucket Electric Company Each d/b/a National Grid D.P.U. 23-150 Exhibit NG-AEB-1 November 16, 2023 H.O. Tassone

# **PRE-FILED DIRECT TESTIMONY**

OF

#### ANN E. BULKLEY

Cost of Capital

Massachusetts Electric Company Nantucket Electric Company Each d/b/a National Grid D.P.U. 23-150 Exhibit NG-AEB-1 November 16, 2023 H.O. Tassone

# TABLE OF CONTENTS

Ĩ.	Introduction		
П.	Purpose of Testimony		
III.	Summary of Analysis and Conclusions4		
IV.	Regulatory Guidelines		
v.	Capita	l Market Conditions	
	Α.	Inflationary Expectations in Current and Projected Capital Market Conditions16	
	B.	The Use of Monetary Policy to Address Inflation	
	C.	The Effect of Inflation and Monetary Policy on Interest Rates and the Investor- Required Return	
	D.	Expected Performance of Utility Stocks and the Investor-Required Return on Utility Investments	
	E.	Conclusion	
VI.	Proxy	Group Selection	
VII.	Cost o	f Equity Estimation	
	Import	ance of Multiple Analytical Approaches	
	Β.	Constant Growth DCF Model	
	С.	Multi-Stage DCF Model	
	D.	Capital Asset Pricing Model	
	E.	BYRP Analysis	
VIII	. Regula	atory and Business Risks	
	A.	Multi-Year Rate Plan	
	<b>B</b> .	Capital Expenditures	
	C.	Performance-Based Ratemaking, Operating Cost Only	
	D.	Regulatory Framework	

Massachusetts Electric Company Nantucket Electric Company Each d/b/a National Grid D.P.U. 23-150 Exhibit NG-AEB-1 November 16, 2023 H.O. Tassone

IX.	Capital Structure, Cost of Debt and Preferred Equity, and Proposed Treatment of Interest	
	Rates for Long-Term Debt Issuance	69
	A. Capital Structure	69
	Cost of Debt and Preferred Equity	75
	Proposed Treatment of Interest Expense for New Long-Term Debt Issuances	76
X.	Conclusions and Recommendations	79

Massachusetts Electric Company Nantucket Electric Company Each d/b/a National Grid D.P.U. 23-150 Exhibit NG-AEB-1 November 16, 2023 Page 1 of 1 H.O. Tassone

# **DIRECT TESTIMONY OF ANN E. BULKLEY**

#### 1 I. Introduction

- 2 Q. Please state your name, affiliation, and business address.
- A. My name is Ann E. Bulkley. I am a Principal at The Brattle Group ("Brattle"). My
  business address is One Beacon Street, Suite 2600, Boston, Massachusetts 02108.

#### 5 Q. On whose behalf are you submitting this testimony?

A. I am submitting this pre-filed direct testimony before the Massachusetts Department of
Public Utilities (the "Department") on behalf of Massachusetts Electric Company ("Mass.
Electric") and Nantucket Electric Company ("Nantucket Electric") each d/b/a National
Grid (together, "National Grid" or the "Company").

#### 10 Q. Please describe your education and experience.

11 A. I hold a Bachelor of Art degree in Economics and Finance from Simmons College and a 12 Master of Art degree in Economics from Boston University, with over 25 years of 13 experience consulting to the energy industry. I have advised numerous energy and utility 14 clients on a wide range of financial and economic issues, with primary concentrations in 15 valuation and utility rate matters. Many of these assignments have included the 16 determination of the cost of capital for valuation and ratemaking purposes. My resume and 17 a summary of testimony that I have filed in other proceedings are provided in Appendix A.

18

Massachusetts Electric Company Nantucket Electric Company Each d/b/a National Grid D.P.U. 23-150 Exhibit NG-AEB-1 November 16, 2023 Page 2 of 81 H.O. Tassone

#### 2 Q. What is the purpose of your pre-filed direct testimony? 3 The purpose of my direct testimony is to present evidence and provide a recommendation Α. 4 regarding the appropriate return on equity ("ROE") for the Company. I also assess the reasonableness of the Company's projected capital structure and cost of debt. Lastly, I 5 discuss the Company's proposal for exogenous treatment of the recovery of the Company's 6 7 long-term debt costs given that interest rates have been volatile and are unpredictable over the five-year term of the proposed rate plan. 8

]

11.

**Purpose of Testimony** 

# 9 Q. Briefly, what are the analyses you have conducted and what factors have you considered that support your recommended ROE for the Company in this proceeding?

I have estimated the Company's cost of equity by applying traditional estimation 12 Α. methodologies to a proxy group of comparable utilities, including the constant growth and 13 the multi-stage forms of the Discounted Cash Flow ("DCF") model, the Capital Asset 14 Pricing Model ("CAPM"), the Empirical Capital Asset Pricing Model ("ECAPM"), and 15 the Bond Yield Plus Risk Premium ("BYRP") analysis. My recommendation also takes 16 into consideration the Company's Comprehensive Performance and Investment Plan ("CPI 17 Plan") and the components of the Company's proposed multi-year rate plan, as presented 18 19 in the pre-filed direct testimony of the CPI Plan Panel at Exhibit NG-CPIP-1. I also considered the Company's proposed capital structure as compared to the capital structures 20 of the utility operating companies of the proxy group. While I do not make specific 21

Massachusetts Electric Company Nantucket Electric Company Each d/b/a National Grid D.P.U. 23-150 Exhibit NG-AEB-1 November 16, 2023 Page 3 of 81 H.O. Tassone

1		adjustments to my ROE recommendation for these factors, I did consider them in the
2		aggregate when determining where my recommended ROE falls within the range of the
3		analytical results.
4	Q.	How is the remainder of your testimony organized?
5	Α.	The remainder of my testimony is organized as follows:
6		• Section III provides a summary of my analyses and conclusions.
7 8		• Section IV reviews the regulatory guidelines pertinent to the development of the cost of capital.
9 10		• Section V discusses current and projected capital market conditions and the effect of those conditions on the cost of equity.
11		• Section VI explains my selection of the proxy group for the Company.
12 13		• Section VII describes my cost of equity estimates and the analytical basis for my recommendation of the appropriate ROE.
14 15 16		• Section VIII provides a discussion of specific regulatory, business, and financial risks that have a direct bearing on the ROE to be authorized for the Company in this case.
17 18 19 20		• Section IX provides an assessment of the reasonableness of the Company's proposed capital structure relative to the proxy group, the Company's proposed cost of debt and preferred equity, and the Company's proposed treatment of its long-term debt during the term of the rate plan.
21		• Section X presents my conclusions and recommendations.
22	Q.	Are you sponsoring any exhibits as part of your testimony in this case?
23	А.	Yes. My analyses and recommendations are supported by the data presented in Exhibits
24		NG-AEB-2 through NG-AEB-18, which were prepared by me or under my direction:
25		• Exhibit NG-AEB-2: Summary of Cost of Equity Model Results
26		• Exhibit NG-AEB-3: Constant Growth DCF

Massachusetts Electric Company Nantucket Electric Company Each d/b/a National Grid D.P.U. 23-150 Exhibit NG-AEB-1 November 16, 2023 Page 4 of 81 H.O. Tassone

1		Exhibit NG-AEB-4: Multi-Stage DCF
2		Exhibit NG-AEB-5: Long-term GDP Growth Rate
3		Exhibit NG-AEB-6: CAPM and ECAPM
4		Exhibit NG-AEB-7: Long Term Beta
5		Exhibit NG-AEB-8: Market Return
6		Exhibit NG-AEB-9: Department Market Return
7		• Exhibit NG-AEB-10: Weighted Market Risk Premium
8		Exhibit NG-AEB-11: BYRP Analysis
9		Exhibit NG-AEB-12: Capital Expenditures
10		Exhibit NG-AEB-13: Capital Tracking Mechanisms
11		Exhibit NG-AEB-14: RRA Regulatory Rankings
12		Exhibit NG-AEB-15: Capital Structure
13		• Exhibit NG-AEB-16: Cost of Debt
14		• Exhibit NG-AEB-17: Overall Rate of Return
15	111.	Summary of Analysis and Conclusions
16 17	Q.	What are the key factors considered in your analyses and upon which you base your recommended ROE?
18	Α.	The key factors that I considered in my cost of equity analyses and recommended ROE for
19		the Company in this proceeding are:
20 21 22 23		• The United States Supreme Court's <u>Hope</u> and <u>Bluefield</u> decisions <sup>1</sup> that established the standards for determining a fair and reasonable authorized ROE for public utilities, including consistency of the allowed return with the returns of other businesses having similar risk, adequacy of the return to provide access to capital

<sup>&</sup>lt;sup>1</sup> <u>Federal Power Commission v. Hope Natural Gas Co.</u>, 320 U.S. 591 (1944) ("<u>Hope</u>"); <u>Bluefield Waterworks</u> <u>& Improvement Co., v. Public Service Commission of West Virginia</u>, 262 U.S. 679 (1923) ("<u>Bluefield</u>").